

Scholars' Mine

Masters Theses

Student Theses and Dissertations

Fall 2011

Comprehensive evaluation of a computer based learning system

Arun Sharma Chintalapati

Follow this and additional works at: https://scholarsmine.mst.edu/masters_theses

Part of the Computer Sciences Commons Department:

Recommended Citation

Chintalapati, Arun Sharma, "Comprehensive evaluation of a computer based learning system" (2011). *Masters Theses.* 4132. https://scholarsmine.mst.edu/masters_theses/4132

This thesis is brought to you by Scholars' Mine, a service of the Missouri S&T Library and Learning Resources. This work is protected by U. S. Copyright Law. Unauthorized use including reproduction for redistribution requires the permission of the copyright holder. For more information, please contact scholarsmine@mst.edu.

COMPREHENSIVE EVALUATION OF A COMPUTER BASED LEARNING SYSTEM

by

ARUN SHARMA CHINTALAPATI

A THESIS

Presented to the Faculty of the Graduate School of the

MISSOURI UNIVERSITY OF SCIENCE AND TECHNOLOGY

In Partial Fulfillment of the Requirements for the Degree

MASTER OF SCIENCE IN INFORMATION SCIENCE & TECHNOLOGY

2011

Approved by

Dr Hong Sheng, Advisor Dr Richard Hall Dr Robert Landers

©2011 Arun Sharma Chintalapati All Rights Reserved

ABSTRACT

This thesis presents the results of a comprehensive multi-method evaluation that was conducted on the Rapid Development System (RDS), a computer-based learning environment. The development of the project was funded by a National Science Foundation grant. Prior to this evaluation, the RDS had gone through two iterations.

A pre/post evaluation was conducted using questionnaires and eye tracking. Multi-method evaluation was employed to help triangulate the results and to provide additional insights. Data from the eye tracking was used to study the influence of the RDS interface on the attention and the cognitive workload of students with respect to the learning styles (visual/verbal).

From the results, a significant improvement was noticed in users' interest levels toward the subject after using RDS. Students reported a significant increase in knowledge after using the system. Minor usability issues were reported using the qualitative data from the eye tracking. We also found differences between visual learners and verbal learners with respect to attention, cognitive workload, perceived ease of use and perceived learning outcome.

Overall, this study found that RDS was well received by the participants and could be used as an effective learning tool in teaching the concepts of control design. Results from this study can help guide the design of computer based learning systems.

ACKNOWLEDGMENTS

First, I would like to thank my advisor, Dr. Hong Sheng, for giving me an opportunity to work under her. I am grateful to her for giving me this project, which has helped me grow both professionally and personally. Her continuous motivation throughout this project helped to bring out the best in me.

I would also like to thank my other committee members for their valuable guidance throughout the project, especially Dr. Richard Hall.

I would like to thank Dr. Robert Landers without whom this project would not be possible. Thanks to Dr. Landers' vision, the budding control engineers of the future now have a modern and motivating system to learn control design.

Finally, I would also like to thank my family members and friends, especially my parents including Anusha, for all the sacrifices they have made for me, for putting up with me throughout this project, and for being available when I needed them.

TABLE OF CONTENTS

Р	a	g	e
•	u	ь	~

	-
ABSTRACT	iii
ACKNOWLEDGMENTS	iv
LIST OF ILLUSTRATIONS	. viii
LIST OF TABLES	ix
SECTION	
1. INTRODUCTION	1
1.1. RATIONALE	1
1.2. RESEARCH OBJECTIVES	2
1.3. THESIS OVERVIEW	3
2. LITERATURE REVIEW	4
2.1. THE COMPUTER BASED LEARNING ENVIRONMENT (CBLE)	4
2.1.1. Animation	6
2.2. LEARNING STYLES	8
2.2.1. The Felder-Silverman Model	9
2.3.USABILITY TESTING	11
2.3.1. Eye tracking	13
2.3.1.1. Eye movement and attention	13
2.3.1.2. Eye tracking technology	16
2.3.2. Using eye movements to evalute interface usability and cognition	19
2.3.2.1. Visualizing eye tracking data	20
2.3.2.2. Quantitative eye tracking data	21
2.3.2.3. Previous eye tracking research related to learning	21
2.3.3. Cognitive workload	23
2.3.3.1. Pupil dilation	24
2.4. PERCEIVED LEARNING OUTCOME	26
3. RESEARCH MODEL AND THEORY	28
3.1. SECTION OVERVIEW	28
3.2. TECHNOLOGY ACCEPTANCE MODEL	28

3.3. LEARNING STYLES	30
3.4. PERCEIVED LEARNING OUTCOME	30
3.5. RESEARCH MODEL	31
4. PROJECT BACKGROUND	32
4.1. OVERVIEW	32
4.2. RDS PHASE II	32
5. DATA COLLECTION	36
5.1. OVERVIEW	36
5.2. PARTICIPANTS	36
5.3. RESEARCH PROCEDURE	36
5.4. INSTRUMENT	37
5.5. DATA ANALYSIS	39
6. QUESTIONNAIRE DATA RESULTS	41
6.1. SECTION OVERVIEW	41
6.2. DEMOGRAPHICS	41
6.3. TASK PERFORMANCE	41
6.4. QUANTITATIVE ANALYSIS	42
6.4.1. Learning styles	42
6.4.2. Interest in science	42
6.4.3. Perceived learning outcome	44
6.4.4. Knowledge gained	46
6.4.5. Technology acceptance constructs (TAM)	48
7. EYE TRACKING RESULTS	51
7.1. SECTION OVERVIEW	51
7.2. DEMOGRAPHICS	51
7.3. RESULTS	51
7.3.1. Overview	51
7.3.2. Qualitative analysis	52
7.3.3. Quantitative analysis	58
7.3.3.1. Effect of learning style on attention towards animation	58
7.3.3.2. Effect of learning style on cognitive workload	59

•

7.3.3.3. Effect of learning style on perceived ease of use	. 59
7.3.3.4. Differences in perceived learning outcome	. 60
8. DISCUSSION AND IMPLICATIONS	. 61
8.1. DISCUSSION	. 61
8.2. IMPLICATIONS	. 63
9. CONCLUSION	. 65
APPENDICES	
A. PRE QUESTIONNAIRE	. 67
B. POST QUESTIONNAIRE	. 71
REFERENCES	. 75
VITA	. 91

vii

LIST OF ILLUSTRATIONS

Figu	re	Page
2.1.	Cross-section of the human eye	
2.2.	Different regions of human vision	
2.3.	Complete scan path with fixation and saccades	
2.4.	Corneal reflections showing glint (bright-pupil effect)	
3.1.	Research Model	
4.1.	The RDS Phase I main interface	
4.2.	The RDS Phase II main interface	
4.3.	The Jog function interface	
4.4.	The simulate function	
5.1.	Area of interest: animation	
6.1.	Learning styles	
6.2.	Interest levels in science	
6.3.	Comparison between learning through RDS, lecture, and notes	
6.4.	Knowledge gained before and after using RDS	
6.5.	Technology acceptance constructs	
7.1.	User notices the Jog button	
7.2.	Gaze plot shows user did not notice pop up message	
7.3.	Heat map shows users did not notice pop up message	
7.4.	Gaze plot shows user did not notice pop up dialog box	
7.5.	Heat map shows users did not notice pop up dialog box	
7.6.	Rapid saccade movements	
7.7.	Heat map of Jog function	57
7.8.	Heat map of help menu	

LIST OF TABLES

Table	Page
6.1. Task performance	41
6.2. Paired samples t-test	43
6.3. Multivariate test results	45
6.4. One way ANOVA results for perceived learning outcome	46
6.5. Paired samples t-test for perceived knowledge gained	47
6.6. Regression model summary	49
6.7. Coefficients of regression	49
6.8. Results of correlation	50
7.1. ANOVA for attention towards animation	58
7.2. ANOVA for cognitive workload	59
7.3. ANOVA for perceived ease of use	60
7.4. ANOVA for perceived learning outcome	60

1. INTRODUCTION

1.1. RATIONALE

E-learning broadly encompasses all forms of electronic teaching (including computer-supported and computer-based) and learning and is becoming more popular every day. In 2007, the American e-learning market was valued at 17.5 billion dollars. This number was expected to more than double to 52.6 billion dollars in 2010 (Kopf, 2007).

One area in e-learning that has seen explosive growth is the use of computerbased learning environments (CBLEs) to aid educators in teaching concepts. The use of CBLEs is becoming more prevalent with the high penetration of computers in the education sector. These environments have become ubiquitous in the classroom to help students learn difficult and challenging topics (Azvedo, 2005; Graesser et al., 2005). Recent advances in computer hardware and software and decreases in the costs of computer hardware have helped to boost the penetration of these learning environments in engineering education to help students understand key engineering concepts (Cheok et al., 1991). In the engineering curriculum, CBLEs help to strengthen course delivery and provide laboratory innovation (Cheok et al., 1991).

The advent of computer-based learning environments coupled with high resolution color graphic capabilities have enabled the development of new technologies such as animation which, when used as part of the learning environment, help to improve conceptual understanding and the appeal of the system. These items enhance the quality of a student's learning process by accommodating the unique learning styles of students (Carlson & Sullivan, 1998).

Recent research in learning styles calls for more testing to understand how learning styles affect the learning outcome and to shed more light on understanding if learning styles cause the software to be perceived differently (Mayer, 2009). This testing can be accomplished by means of a thorough usability test in which the end user interacts with a system to provide valuable indicators on the performance of the interface.

Usability testing plays a central role in the development of interactive educational software. A user-centered design is quintessential for promoting the usage of such software. Usability problems in educational software programs can cause disturbance within the learning process by distracting the learner's attention from the learning task and, consequently, increasing the cognitive workload on the learner (Domagk et al., 2004).

Research in the area of e-learning has shown that progress in this area has been slow due to problems related to the poor interface designs of the learning systems (Zaharias, 2005). Furthermore, despite the importance of including usability evaluations in the design of learning systems, research in this area has indicated that usability evaluations are not frequently used (Sigchi, 2001).

1.2. RESEARCH OBJECTIVES

In this thesis, we build on the previous research done in this area where a comprehensive usability framework (Jain et al., 2009) was developed to evaluate educational technologies. We utilized this framework in evaluating a computer-based learning system called the Rapid Development System (RDS), which was designed for teaching control design concepts. The main objective of the evaluation was to test if users

would accept the system as a learning tool. The second objective was to understand how an interface that included animation as a significant part of the system influenced the learning outcome between two groups: (a) visual learners and (b) verbal learners.

1.3. T ESIS OVERVIEW

Section I covered the motive of our research, the research objective and the gap in the literature that we need to address. Section II, the literature review, examines the previous literature on the concepts used in our research.

Section III describes the research model with constructs adapted from previous literature. Section IV describes the overview of the project.

Section V discusses our approach and the procedures that were followed in conducting the study, the approach taken and materials used in this research. Section VI presents and explains the findings of the questionnaire data. Section VII presents and explains findings from the eye tracking data.

Finally, Section VIII summarizes the findings and uses both qualitative and quantitative results to determine the conclusions in section IX.

2. LITERATURE REVIEW

2.1. THE COMPUTER BASED LEARNING ENVIRONMENT (CBLE)

"Learning environments are comprehensive, integrated systems that promote engagement through student-centered activities, including guided presentations, manipulation, and explorations among interrelated learning themes" (Hannafin, 1992). Learning environments that use computer aided information delivery are called computer-based learning environments (CBLEs). These environments are increasingly used to supplement traditional forms of education (Carlson & Sullivan, 1998).

Computer-based learning environments enable individuals with different learning needs and unique learning requirements to obtain a deepener understanding of a subject and to study multiple hierarchies of complexity in an interactive and complementary manner (Hannafin & Land, 1997). CBLEs can boost both the speed and the level of student learning (Horton, 2000; Najjar, 1998). These environments also help to improve the students' confidence and motivation (Klassen et al., 2001). Using an interactive webbased learning program can increase the learning enjoyment level, which, in turn, may increase a student's understanding and his or her information retention (Street & Goodman, 1998).Thus, computer-based learning environments provide a fertile ground that help to enrich thinking and learning.

Understanding complex systems to help solve real world problems is an important part of engineering education. In the recent decade, universities have rapidly embraced CBLEs for providing in-depth real-world laboratory education for engineering disciplines. Engineering is also a practical discipline and a hands-on profession where the "doing" is key (Feisel & Rosa, 2005). The use of hands-on experimentation using either the equipment or tools that simulate this equipment is a key aspect of engineering education (Carlson & Sullivan, 1998). Previous research has shown that hands-on experimentation in laboratories are an essential supplement to the relatively passive experiences of learning via listening to lectures and reading textbooks (Edelson et al., 1999; Bransford et al., 1990; Lave &Wenger, 1991). Knight et al. (2007) found that courses that were designed to include active hands-on pedagogy improved the retention of students in engineering education.

Currently, computer-based learning environments are being used as an integral part of teaching in the engineering curriculum. These learning environments enable students to emulate many physical and technical processes for the experimental studies of systems that would otherwise be expensive, large, or dangerous to test physically (Feisel & Rosa, 2005). Previous studies have shown that students who learn using computer-based animation environments have a higher understanding of complex concepts and systems compared to students who learn in a traditional learning environment that focuses on verbal explanation (Park, 1994; Reiber, 1991; Tversky et al., 2002). Computer-aided engineering tools make it possible to increase the personal efficiency of problem solving. These tools also allow students to work on realistic problems which contribute to making teaching more realistic and more interesting (Kheir, 1996).

An overview of the design of various computer-based learning systems (Evans & Edwards, 1999; Evans et al., 2004) shows that most of these systems resemble traditional textbooks and lack interactivity (Markwell & Brooks, 2002). This results in learning

through observing as opposed to hands-on learning (Moos & Azevedo, 2009). This also results in students using strategies such as memorizing the paths and completing the tasks on the computer-based learning systems without gaining knowledge. In view of this, Mayer (2009) recommended the use of a learner-centered approach (similar to a usercentered approach) in developing these systems where the goal is to design the system in such a way that it aids the processing of information.

One method of designing CBLEs that aid the processing of information is by building features that support different learning styles. Learners attend differentially to selected stimuli and invest their effort accordingly (Hannafin, 1992). New technologies have made new presentation formats available (e.g. animation, narration, and cueing) (Bannert, 2002), which can be built into the learning systems to support the unique learning styles of students. Building these features into an interface allows a student's attention and effort to be selectively directed. These methods allow learners to be engaged towards the content and to learn in ways that are uniquely suited to the individual's perception.

2.1.1. Animation. Animation is defined as a visual representation that "generates a series of frames, so that each frame appears as an alternation of the previous one" (Betrantcourt, 2005). Animation can be used to present events (e.g. motions) that change over time. Animation helps learners visualize and create internal representations of how the system works. As a dynamic representation, animation can help learners understand complex processes in an explicit manner (Schnotz & Lowe, 2003).

Animation involves a complex interplay between top-down and bottom-up processes (Kriz & Hegarty, 2007). In order to grasp the information from the animation, learners must be able to extract thematically relevant information (Lowe, 2003).

An important area where animation is being used is in the simulation of control systems. Animation reduces the need of expensive equipment and increases safety. The use of the animated visualization of dynamic system motion can help enhance the perception and understanding of a system that is subjected to simulation (Kheir et al., 1996). At the same time, animation brings the simulations to life and helps to portray a sense of reality (Kheir et al., 1996). Previous research in the area of control systems has shown that students' conceptual understanding of simulation results and control design concepts can be enhanced significantly by their interaction with real time simulation and animation (Cheok et al., 1991; Cheok et al., 1993; Cheok & Huang, 1992; Cheok & Kheir, 1993).

An extension of this research topic involves the study of the role of individual learning characteristics in influencing the use of such systems (Plass et al., 1998; Riding, 2001). Thus, one research direction is to understand which factors contribute to a higher adaptability and drive the increased usage of CBLEs built with animation (Hoffler et al., 2010). Some of the factors that have been studied are prior knowledge and spatial ability (Hoffler et al., 2006; Huk, 2006; Isaak & Just, 1995). However, very little research has investigated if visual learners perceive their interaction with the learning environment differently than verbal learners do. Additionally, there is also a need to examine if the use of such systems can place a cognitive workload on individual learners with respect to their preference in learning (Schnotz et al., 1999).

In the next section, we discuss learning styles that are unique to every individual and are a useful classification to help drive design decisions.

2.2. LEARNING STYLES

In a learner-centered teaching, pedagogy should be based on learners' needs rather than on teachers' or institutions' needs and should be compatible with the use of information and communication technology. In education, research has aimed at understanding the individual differences in learning processes. These differences are called "learning styles."

Learning styles are defined as "characteristic cognitive, affective, and psychological behaviors that serve as relatively stable indicators of how learners perceive, interact with, and respond to the learning environment" (Keefe, 1979).Students are characterized by their preference of how they want to learn and how they operate on the information they learned (Corno & Snow, 1986).Previous research has shown that students are characterized by a variety of learning styles (Corno & Snow, 1986; Schmeck, 1988).

There are many different types of learning style models including Honey and Mumford's model (1984), Kolb's model (1984), and Felder and Silverman's model (1988). Each model classifies learning styles and provides different descriptions for the classification. In our research study, we used the Felder-Silverman model, a learning style model that is often used to study advanced technology-enhanced learning (Graf, 2007).

The Felder-Silverman model of learning style classification was chosen to classify students since it provides more depth in its classification and since it has a larger number of groups compared to other methods (Graf et al., 2007). Carver et al. (1999) stated that, "The Felder-Silverman Model is most appropriate for hypermedia courseware." Kuljis and Liu (2005) also suggested the Felder-Silverman model as the most appropriate model based on their research in which they compared various learning style models with respect to applications in e-learning and web-based learning systems.

2.2.1. The Felder-Silverman Model. The Felder-Silverman model (Felder & Silverman, 1988) is one of the most prominent models used to categorize engineering students' learning styles (Montgomery, 1995). In this model, students are categorized based on their learning preferences and on how they act on the learning.

- The type of information the student prefer to perceive: sensory (such as sights, sounds, and physical sensations) or intuitive (such as memory, insight, and feelings).
- The preferred sensory mode through which the information can be best perceived: visual (through pictures, animations, sounds, etc.) or verbal (through written and/or spoken instructions).
- The student's preferred method to process the information: actively (i.e. via discussions or physical engagement) or reflectively (i.e. via introspection).
- 4) The way in which the student progresses towards understanding: sequentially (in incremental progression) or globally (understanding the big picture or in big jumps).

Visual learners remember best what they see. They tend to find diagrams, sketches, schematics, photographs, flow charts, or any other visual representation of course material that is primarily verbal and very useful to learn. Verbal learners, on the other hand, learn through words through written explanations and/or spoken explanations. They write summaries or outlines of course material in their own words, work in groups to have a more effective learning experience, gain an understanding of material by hearing classmates' explanations, and learn even more when they do the explaining.

In engineering education, studies have attempted to relate learning style to learning outcomes in order to make engineering education more effective. The results show that learning style theory is a potential tool for guiding the design and improvement of courses and for helping students to improve their individual performance (Carver et al., 2005). Eder and Hubka (2005) presented various learning styles as one element in their model of educational design as a transformational process that takes a student from an input state to an output state. Each student has a unique combination of knowledge, skills, attitudes, values, and learning style (Eder & Hubka, 2005). In order to increase a student's success in engineering education, one must understand the student's individual learning style and provide instructional methods and environments accordingly (Kyun et al., 2009). Researchers in applied fields have found that an individual's learning style can be a better predictor of his or her success in a particular situation than general intelligence or situational factors are (Kyun et al., 2009).

One way to address diverse learning styles is to use computer-based learning. In addition, an awareness of the pedagogical needs of various learning styles can result in the development of a more effective computer-based learning system. By assessing the learning style of students, interfaces for education software should be designed to cater to a variety of different learning styles (Kramer-Koehler et al., 1995). A study conducted by Azvedo et al. (2004) proved that, while the use of CBLE resulted in a large gain in the conceptual understanding in one group of students, it resulted in very little or no gain in another group of students. For this study, students were divided into two groups based on the shift in their mental models in understanding: (a) high jumpers (higher conceptual learners) and (b) low jumpers (low conceptual learners) (Azvedo et al., 2004). Educational researchers and instruction designers suggest that one of the causes for this perplexing difference in the performance of individuals could be learning styles (Streufert & Nogami, 1989; Hayes & Allison, 1994).

Learning experience, described as the transaction that takes place between a learner and the instruction environment, is different for each learner (Parrish, 2009). It is not enough to provide students access to different tools and/or learning environments without taking the learner into consideration (Bates & Leary, 2001). Therefore, these learning environments must be designed with consideration of the end user. In order to design such a system, a user-centered design process must be followed. One of the hallmarks of this approach is to evaluate the system by following a distinct interactive system evaluation using the end users as participants (Granic, 2008). This is called a "usability evaluation."

2.3. USABILITY TESTING

According to the International Organization for Standardization (Bevan, 2001) usability is defined as "the extent to which a product can be used by specified users to

achieve specified goals with effectiveness, efficiency and satisfaction in a specified context of use." Usability is an important aspect of an interactive learning system. A higher usability in a learning system would facilitate an increase in user satisfaction and user acceptance that would result in the increased adoption of the system (Danao, 2010).

Research in the area of human-computer interaction has provided numerous principles and design guidelines to help designers build usable systems. However, following the design guidelines alone is not a substitute for a distinct system-wide interactive evaluation in order to ensure usability (Granic, 2008). In order to build highly usable systems, an iterative process that involves design, evaluation, and redesign is essential (Hartson et al., 2003). This approach takes into consideration the users' natural behavior and includes any constraints on the ability to learn and perform so that the interfaces are more intuitive, easier to use, easier to learn, and free of performance errors.

Usability testing requires representative users to work on typical tasks using the system or the prototype (Ardito et al., 2006). The results from the evaluation provide evaluators with an understanding of how the user interface supports the user while carrying out the tasks.

Over the years, many analytical usability evaluation methods have been developed. Each method addresses a particular type of usability issues. Some of the popular methods of evaluation are the Heuristic Evaluation method (Nielsen, 1994) and the Cognitive Walkthrough method (Cuomo &Bowen, 1994; Desurvire, 1994; Dutt et al., 1994). Another method that is being used frequently to study and identify usability issues is the eye tracking method (Poole & Ball, 2006). **2.3.1. Eye tracking.** Eye Tracking is a method in which eye movement data over a visual stimulus is collected. This method allows researchers to ascertain the position of the eyes as they move over a visual stimulus. Thus it provides researchers with information on the distribution of visual (overt) attention over different objects (e.g. words in a sentence or different areas of GUI) in term of what they see, for how long, and in what order (Scheiter & Van Gog, 2009). Previous research in the area of visual attention has suggested that what is being currently fixated at is an indication to what is being processed in the mind (also called eye-mind hypothesis) (Just & Carpenter, 1980). Eye tracking can thus provide researchers with a dynamic trace of user's attention over a visual display (Poole & Ball, 2005).

In order to understand how eye movements can help in quantifying users' attention towards the interface elements it is important that we understand the relationship between eye movements and cognitive processes.

2.3.1.1. Eye movement and attention. Previous research in the area of neuroscience and psychology has argued that studying attention independently of eye movements is misleading and misguided (Findlay & Gilchrist, 2003). Jacob (1995) suggests that a complete theory of visual processing needs to include an account of eye movements. In order to understand how eye movement and attention work it is imperative that we understand the physiology of the eye.

The human eye is composed of three layers enclosed by three transparent structures. The outermost layer consists of cornea and sclera. The middle layer consists of choroid, cillary body and iris (Hammoud, 2008). And the innermost layer consists of the retina. The figure 2.1 below shows the cross section of the human eye.



Figure 2.1. Cross-section of the human eye (Hammoud, 2008)

When an individual looks at an object, an image of the object falls on the retina. The retina is composed of special light sensitive cells called cones and rods that convert the light signals to optical information and transmit it to the brain via the optical nerve. However, the distribution of these cells is not even. The center of the retina called the fovea (foveal vision) has higher densely packed cells that gradually become sparser towards the periphery of the retina (parafoveal vision) (Pashler, 1998). The figure 2.2 below shows the different regions of human vision.

The fovea covers approximately one degree field of view, that is, a one-degree angle with its vertex at the eye, extending outward into space. Thus the fovea provides much higher acuity vision than the surrounding areas. The eye does not move smoothly over the visual field (Duhowski, 2003). Eye movements are made to reorient the eye when visual input from a particular location is of special importance so that the object of interest falls upon the fovea and the highest level of detail can be extracted (Pashler, 1998). This movement during which the eye focuses on an object is called a fixation.



Figure 2.2. Different regions of human vision (Hammoud, 2008)

The duration of each fixation is in the range of 250-300 milliseconds (Salvucci & Goldberg, 2000). In order to get to the fixation or move to the next fixation the eye goes through a movement termed "saccade" (Pashler, 1998). The saccade is used to orient the eyeball so that the desired portion of the visual scene falls upon the fovea (Pashler, 1998).

Saccades are rapid, ballistic eye movements (high acceleration and deceleration rates) that last 30-120 milliseconds from planning to execution (Palmer, 1999). Since the saccade is ballistic, its destination must be selected before movement begins; since the destination typically lies outside the fovea it must be selected by lower acuity peripheral vision (Pashler, 1998). If an object that might attract a saccade suddenly appears in peripheral vision, there is a 100-300 milliseconds delay before the saccade occurs. No information is perceived during a saccade as the perception is inhibited to prevent the viewer seeing a blur. Information can only be perceived when the eyes are relatively still i.e. during a fixation (Duhowski, 2003). Thus a complete scan path consists of fixation

during which information is encoded and saccades to move to then next fixation. The figure 2.3 below shows the complete scan path with fixation and saccades.



Figure 2.3. Complete scan path with fixation and saccades (Hammoud, 2008)

Tracking these eye movements can provide researchers with valuable clues on how users process information. Active vision ensures the availability of high quality visual information to support perceptual and cognitive processing as well as behavioral activity, and help in simplifying a variety of complex interactions both visual and cognitive in nature (Ballard, 1976; Ballard et al, 1997; Churchland et al., 1994).

In the next part we briefly discuss some of the eye tracking technologies and how modern eye trackers work.

2.3.1.2. Eye tracking technology. A variety of eye tracking technologies have been developed since the first reported use of the methodology (Rayner & Pollatsek,

1989). In general, eye tracking methodologies can be classified into two types: those that measure the position of the eye relative to the head and those that measure the orientation of the eye in space also called, "point of regard" (Young & Sheena, 1975). From a technological stand point eye tracking techniques can be broadly classified into four different types: electrooculography (EOG), scleral contact lens/search coil, photo-oculography or video-oculography (VOG) and video based combined pupil and corneal reflection.

Electrooculography relies on electrodes placed around the eye to measure differences in electric potential to detect eye movements (Duchowski, 2003). This method was widely used previously and still in use. The sclera contact lens/search coil is one of the most accurate methodologies to measure eye movements (Duchowski, 2003). However, this method is also the most intrusive. It involves the wearing of specialized lenses that has a metal coil embedded around the edge of the lens; eye movements are measured by fluctuations in an electromagnetic field when the metal coil moves along with the eyes (Duchowski, 2003). The third technique involves the measurement of various distinguishable features of the eye such as the apparent shape of the pupil, corneal reflections and the iris-sclera boundary using video camera (Duchowski, 2003). Nowadays most of the modern eye trackers utilize the pupil centered corneal reflection to measure point of regard (Goldberg & Wichansky, 2003).

Modern eye trackers consist of specialized chassis that houses standard LCD monitor along with an eye-tracking unit located at the bottom (Poole & Ball, 2005). The eye tracking unit contains two sets of IR led emitters and IR cameras to track both eyes. These are called binocular systems as they track both the eyes. The eye tracker also

houses a specialized microcomputer that runs sophisticated image processing algorithms to identify relevant features, including the eyes and the corneal reflection patterns (Duhowski, 2003).

The overall setup consists of an eye tracker connected to a dedicated desktop/laptop computer running specialized software to create tests (setting the stimulus), record the eye movements and help in data analysis. During tracking, the eye tracker fires the near infrared diodes to generate reflection patterns on the corneas of the eyes of the user (Poole & Ball, 2005). These corneal reflections are also called as purkinje reflections (Duhowski, 2003). The light from the LED enters the retina and large portion of it is reflected back and in the process illuminates the pupil (called Bright pupil effect see figure) (Poole & Ball, 2005). The corneal reflections from the infrared light appear as a small yet sharp glint as shown in figure 2.2 below.

An infrared camera collects these reflection patterns, together with other visual information about the person. Sophisticated image processing algorithms in the software identify relevant features, including the eyes, corneal reflection patterns and the center of the retina (Duhowski, 2003). Complex mathematics is used to calculate the three-dimensional position of each eyeball, and finally the gaze point on the screen, i.e. where the user is looking. As each individual has unique eye properties that needs to be mapped to the three dimensional eye model built into the eye tracker, a calibration needs to be performed (Duhowski, 2003). This calibration generally uses a 5-point (95% accuracy) system. In the calibration phase individuals are asked to follow a dot on the screen, if the eye fixes for a longer than a certain threshold and is within a certain area, the system

records pupil centered corneal reflection relationship to specific a coordinate (x, y) on the screen (Poole & Ball, 2005).



Figure 2.4. Corneal reflections showing glint (bright-pupil effect)

In the next section we will discuss relevant literature on the role of eye tracking in studying learning and finding usability issues.

2.3.2. Using eye movements to evaluate interface usability and cognition. The use of eye movements has tremendously befitted the psychology community as they can provide an insight into problem solving, reasoning, mental imagery, and search strategies (e.g., Ball et al., 2003; Just & Carpenter, 1976; Yoon & Narayanan, 2004, Zelinsky & Sheinberg, 1995). The use of eye movements in studying cognitive processes has been documented by Rayner (1998). One of the most important areas where eye tracking could be applied is in revealing moment to moment processing activities (Rayner, 1998). An important study in this area is the eye-mind hypothesis (Just & Carpenter, 1980) which suggests a close link between the gaze and attention. It suggests that there is a relationship between what is currently looked at and the current visual processing.

The "eye-mind" hypothesis indicates what a person observes is assumed to indicate the thought "on top of the stack" of cognitive process (Just & Carpenter, 1980). This means that eye movement recordings can provide researchers with a trace of the user's attention on an interface. Other eye tracking measures such as fixations (stationary eye movements) can indicate the processing time on the area that was fixated (Hyönä, 2009).

2.3.2.1. Visualizing eye tracking data. Eye tracking data can be visualized using two popular formats called gaze plots and heat maps (Goldberg & Wichansky, 2000).

The Gaze Plot visualization shows the sequence and position of fixations (dots) on a static media, (e.g. an image or a scene). The size of the dots indicates the fixation duration and the numbers in the dots represent the order of the fixations. Gaze plots can be used to illustrate the gaze pattern of a single test participant throughout the entire eye tracking session, or of several participants in a short time interval (Duhowski, 2003).

Heat maps on the other hand are aggregated over multiple participants. These can be of great value when creating reports, papers or presentations, as they help you to summarize large quantities of data in an intuitive way. A heat map uses different colors to show the number of fixations participants made in certain areas of the image or for how long they fixated within that area (Duhowski, 2003). Red usually indicates the highest number of fixations or the longest time, and green the least, with varying levels in between.

For relative duration heat maps which were used in this paper, the duration of each fixation is divided first by the media viewing time and then added. Once all the fixations values have been added together, color values are added to all the points, with the warmest color (red) representing the highest value.

2.3.2.2. Quantitative eye tracking data. In practice, the process of inferring useful information from eye-movement recordings involves the defining of "areas of interest" over certain parts of a display or interface under evaluation, and analyzing the eye movements which fall within such areas (Poole & Ball, 2005). In this way, the visibility, meaningfulness and placement of specific interface elements can be objectively evaluated and the resulting findings can be used to improve the design of the interface (Goldberg & Kotval, 1999).

One important measure that is obtained from the eye tracking data is the total fixation duration. This total fixation duration is one of the most popular measures used by researchers to measure a user's attention towards an interface element in the area of multimedia learning with graphics (Mayer, 2010). The total fixation time on relevant areas reveals the perceptual processing during learning. A study by Schmidt-Weigand et al. (2010) used fixation time on visualization and text and found that learners spent more time viewing visualizations with spoken text than with written text. To rephrase, this metric measures the sum of all fixation durations that are within the area of interest. The total fixation time on the relevant areas reveals the perceptual processing during learning.

2.3.2.3. Previous eye-tracking research related to learning. Previous research in the area of learning using eye tracking has focused primarily on reading research (Hyona & Niemi, 1990; Just & Carpenter, 1980; Rayner, 1998). Later research focused on the other areas of learning such as through pictures and text and problem solving. Current research has focused on the use of eye movements in evaluating multimedia

based learning (Schmidt-Weigand et al., 2010; Kaakinenn et al., 2002). For example, Boucheix & Lowe (2010) looked at learning through animation and found that visual signals improved the perceived learning outcome.

Hegarty and Just (1993) studied how learners attempt to integrate verbal and pictorial information. Eye tracking data from the study revealed interesting findings. Results showed that the processing of the diagram appeared to be largely text-guided, in that participants first read a text in increments before constructing a spatial mental model based on the pictorial information. An important effect seen by researchers among student learning through multimedia learning is the split-attention effect (Ayres & Sweller, 2005). This effect indicates that two mutually referring but separately presented information sources (text-picture combinations, cf. Mayer, 2005), hampers learning as compared to an integrated presentation format. However, more research is needed to understand how different presentation techniques i.e. mutually representative but presented separately can hamper learning in comparison to presenting information in an integrated approach. It is hypothesized that this could be due to high processing demands imposed by visual search and mental integration, but the exact nature of these processes is unknown, and so is whether these processes are the same for all learners (Scheiter & Van Gog, 2009).

Hannus and Hyońä (1999), Hegarty and Just (1993) conducted research in this area using eye tracking. Hannus and Hyońä (2009) studied how children process illustrated texts. Results from their study showed that children focused on reading the text, and mostly ignored the illustrations and individual cognitive abilities moderated identification of relevant material. Another research analyzed the eye movements of newspaper readers in response to different designs of information graphics and related text (Holsanova et al., 2008). Results from the study suggest that when the text and graphic are placed separately, readers treat the text and graphics as separate units, making fewer attempts to integrate the information from both sources. The results from the study can be an explanation why the split-source formats tends to hamper learning in comparison to integrated formats (Ayres & Sweller, 2005). One reason why split- source formats hamper learning can be attributed to the fact that it is difficult to simulate a proper mental representation (Scheiter & Van Gog, 2009). Holsanova et al. (2008) also showed that arrangement of graphic elements could influence processing of the graphic elements.

Arranging multiple graphical elements in a way that suggests a logical, serial reading path leads to more intensive processing of the graphic and more integrative saccades as opposed to following a radial pattern.

Thus research in this area demonstrates how the presentation of media can affect the visual attention, and by doing so contribute to our understanding of why combining information from texts and graphics is supported by certain designs of multirepresentational environments and hampered by others (Scheiter & Van Gog, 2009). An important extension to this research is to understand if cognitive preferences can affect visual attention towards multi-representational environments (Mayer, 2005).

In the next section we look at the cognitive workload measure which can be measured using the pupil dilation measure from eye tracking.

2.3.3. Cognitive workload. Cognitive workload is a multi-dimensional concept.One way to define it is "the cognitive workload of a task represents the level of

attentional resources required to meet both objective and subjective performance criteria, which may be mediated by task demands, external support and past experience" (Young & Stanton, 2004). This definition is based on the assumption that attentional resources have a finite capacity beyond which any further increase in demand results in performance degradation.

Physiological, performance-based, and self-assessment techniques all have been used to measure cognitive workload (Paas et al., 2003; Farmer & Brownson, 2003; Kobus & Morrison, 1905). The typical performance measures of cognitive workload have included indices such as time to complete tasks, reaction time, correct solutions, memory retrieval time and correctness, time estimation, rate of physical activity and speech, spoken disfluencies, and multimodal integration patterns (Ho &Spence, 2005; Oviatt, 1995; Oviatt et al., 2006; Oviatt et al., 2003; Paas et al., 2003). In contrast, subjective measures of cognitive workload interrupt a person's work and they cannot be collected as real-time measures (Farmer & Brownson, 2003). The more promising physiological measures for assessing cognitive workload, such as brain activity, is reflected in EEGs and the eye monitoring of pupil size (Oviatt, 2005).

2.3.3.1. Pupil dilation. Previous research has suggested that dilation of the pupil is the best single index to measure autonomic indication of effort followed by increase in skin conductance (Colman & Pavio, 1969; Kahneman et al, 1969). An important requirement of a physiological measure of mental effort is that it should be sensitive to both within-task and between task variations (Kahneman, 1973). A perfect measure would be one that would enable the measurement of mental effort that different people invest in a given task. Measurement of the pupil diameter satisfies the first two

requirements by providing a sensitive indication of both between task and within task variations of effort (Goldwater, 1972).

The claim that pupil dilations can indicate cognitive workload was made by Hess and Polt (1964). They observed a striking correspondence between mental arithmetic problems and magnitude of dilation during the solution period. This relationship between cognitive load and pupil dilation was later confirmed in numerous contexts: performing arithmetic calculations (Bradshaw, 1968; Payne et al., 1968); short-term memory tasks of varying load (Kanheman & Beatty, 1966); pitch discriminations of varying difficulties (Kahneman & Beatty, 1967); standardized tests to measure concentration (Bradshaw, 1968a); sentence comprehension (Wright & Kahneman, 1971); paired-associate learning (Colman & Paivio, 1970; Kahneman & Peavler, 1969); imagery tasks using abstract and concrete words (Paivio & Simpson, 1966, 1968; Simpson & Paivio, 1968). In all the situations discussed above, the amount of dilation increased with task demand or difficulty.

There exists a relationship between pupil dilation and attention even in the absence of specific instructions. An example is the research that found observed dilations of the pupil when participant looked at pictures (Libby & Lacey, 1973). Largest dilations occurred during exposure to attention grabbing and interesting pictures (Kahneman, 1973). Pratt (1970) observed that pupil dilations varied with unpredictability of random shapes to which participants were exposed. Thus, it is evident that complex and interesting pictures, like difficult tasks, attract attention and demand a relatively large investment of effort (Kahneman, 1973).

An important requirement of an adequate measure of effort is within-task sensitivity. Many studies have confirmed the suggestions that the size of the pupil at any time during performance reflects participant's momentary involvement in the task (Hess, 1965). This is indeed true as the fidelity of the pupil response permits a second-bysecond analysis of task load and effort. An example is the research done by Kahneman and Beatty (1966), who showed that presentation of each successive digit in a short term memory task, was accompanied by a dilation of the pupil.

The pupil dilation measure is a relatively fast response and major dilations can occur within one second after the presentation of the stimulus depending on how demanding the stimulus is (Beatty & Kahneman, 1966). Beatty and Lucero-Wagoner (2000) identified three important task-evoked papillary responses (TEPRs): (a) mean pupil dilation, (b) peak dilation, and (c) latency to the peak. These measures typically increase as a function of a user's cognitive workload (Beatty & Lucero-Wagoner, 2000). In this study, mean pupil dilation was used for measuring cognitive workload.

2.4. PERCEIVED LEARNING OUTCOME

Perceived learning outcome refers to the perceived knowledge gained by the students through the use of different methods of teaching or studying. A learning outcome includes statements that describe the knowledge, skills, and attitudes that learners should have after successfully completing a learning experience or program.

Methods of teaching or studying typically include reading textbooks, performing experiments in a laboratory setting, and using information technologies. The assessment of learning outcomes often includes quantitative measures for notions such as motivation
to learn, real world applicability, and knowledge or learning awareness. These perceived learning outcomes were adapted from the study of a comprehensive learning system by Hall et al. (2006).

3. RESEARCH MODEL AND THEORY

3.1. SECTION OVERVIEW

The research model (Jain et al., 2009) used in this study is a multi-dimensional model that allowed us to test various aspects such as the user acceptance of the technology. This model was developed by incorporating various constructs from different literature. The proposed model can be used in a full-scale evaluation of IT based learning technologies. It can objectively measure the user's acceptance of the technology, the learning outcome associated with the usage of the technology, and if the technology is designed to accommodate the unique learning styles of students. In this section, we discuss how this robust model was developed by incorporating highly tested constructs from previous literature.

3.2. TECHNOLOGY ACCEPTANCE MODEL

There are many models used in previous studies to study IT user acceptance. Some of them are the Theory of Reasoned Action (TRA) (Fishbein & Ajzen, 1975) which is considered to be the ancestor of all IT acceptance models. Another model is the Technology Acceptance Model (TAM) which was also developed from TRA by incorporating the perceived usefulness and perceived ease of use constructs. The theory of planned behavior (TPB) is another model used to study user acceptance which is also based on TRA and includes the perceived behavioral control construct (Ajzen, 1991). The Unified theory of Acceptance and Use of Technology (UTAUT) is a unified model developed by Venkatesh et al. in 2003 which includes constructs from all other user acceptance models such as TRA, TPB, TAM, etc.

One of the models that have been extensively tested, studied, and used in prior literature for studying the user acceptance of technology is the Technology Acceptance Model (TAM) (Davis, 1989;). TAM has been proposed to specifically explain computer/IT usage behavior and thus was considered apt for this study.

The Technology Acceptance Model suggests that the perceived ease of use of the technology together with the perceived usefulness influences a user's attitude towards the technology and, consequently, his or her intention to use it. The perceived usefulness is defined as "the degree to which a person believes that using a particular system would enhance his or her job performance" and the perceived ease of use of the technology is "the degree to which a person believes that using a particular system would be free of effort"(Davis, 1989). The perceived ease of use and the perceived usefulness of the technology determine the user's attitude towards the technology, which, in turn, affects the users' intention to accept the technology. Attitude is defined as "the favorable or unfavorable feeling towards that particular behavior" (Fishbein & Ajzen, 1975).

The Technology Acceptance Model provides a basis for a practical and effective "user acceptance testing" that predicts the degree of user acceptance of a new system based on self-reported measures from users who have limited exposure to the system (Shneiderman et al., 2006). Since TAM has proven to be an effective model in predicting users' intentions to adopt or use new technology, it can be applied to study the acceptance of relatively new technology. Since, RDS was a relatively new system we used TAM in our research model. Since IT learning systems are specific to a group of users, none of the existing models can reflect the motives of the e-learners, thus requiring addition of more motivational factors specific to the system (Ong et al., 2004). Many variables from the previous models such as perceived ease of use and perceived usefulness have been used to explain the acceptance and the intention to use a system (Ong et al., 2004; Teo, 2009). Other researchers have used additional factors such as facilitating conditions and social conditions (Teo, 2009), perceived playfulness (Moon & Kim, 2001), self-management of learning (Smith et al., 2003), enjoyment (Yi & Hwang, 2003) specific to the system and relevant to the learning procedure in order to explain the intention to use a system.

In our research since we wanted to study the effect of learning styles and learning outcome, these constructs were used along with the TAM and a research framework was thus developed by Jain et al. (2009).

3.3. LEARNING STYLES

The Felder and Silverman (1988) model was added into this study as a moderating variable and was used to classify students based on their preference of learning. This model allowed us to test if the technology was designed to accommodate the unique learning characteristics of the student.

3.4. PERCEIVED LEARNING OUTCOME

The Felder and Silverman (1988) model was further extended by including new variables, such as perceived learning outcome (Hall et al., 2006), which allowed us to

measure the students' learning after using the learning system. The extended model allowed us to measure if students had an increase in their perceived knowledge on the subject after using the system, which is the important goal of the learning system.

3.5. RESEARCH MODEL

A diagram of our research model is shown in Figure 3.1. This model allowed us to evaluate the learning of the student in the subject matter after using the technology objectively. It helped us understand if the technology accommodated the unique learning styles of the students. The research model also helped us test the user acceptance of the technology by evaluating if the students perceived their use of the system to be easy and useful to them.



Figure 3.1. Research Model

4. PROJECT BACKGROUND

4.1. OVERVIEW

The Rapid Development System (RDS) is a hardware/software educational technology that has been under continuous development by the Mechanical and Aerospace Engineering Department at Missouri Science and Technology with funding from the National Science Foundation. The RDS technology was developed to provide hands-on experience to students enrolled in control systems classes that are part of the mechanical engineering curriculum.

4.2. RDS PHASE II

The present research was limited to Phase II A and Phase II B. The first phase, RDS Phase I, was excluded from the research for two reasons. First, the interface used in this phase (see figure 4.1.) did not include the animation element as a part of the interface. Second, the sample size of students who participated in the research in this phase was very small (three participants). Nevertheless, the RDS Phase I was a major breakthrough attempt in providing students with a practical hands-on control laboratory environment that illustrated some of the basic aspects of control engineering, modeling, simulation, and implementation.

RDS Phase II was a major improvement over RDS Phase I. It was re-designed with major improvements such as the inclusion of a completely revamped intuitive graphical user interface (GUI). The overall interface was widened. An interactive help menu was created to provide step-by step help instructions to the user.

pidcontrol	
Node	Controller
Coloris Made	Colori Controllor
ana na	Energine and a strand and a strand and a strand and a strand a st

Figure 4.1. The RDS Phase I main interface.

This menu was made accessible by clicking a prominently visible help button. A second inclusion was a tip box, which provided the user with simple guidelines on the usage. The third inclusion was a feedback box that flashed messages, which guided the user in the initial stages. The fourth major inclusion was an animation that helped visualize the control process (a simulation of the physical Computer Numerical Control (CNC) and provided a more interactive way of learning. The software also provided time-response plots that showed how the system responds to external excitations, which is also an important aspect of learning control systems (Kheir et al., 1996).

RDS Phase II has a graphical user interface with three main modes: (a) simulate, (b) emulate, and (c) implement. In the simulation mode, the student simulates the linear axis system that includes their controller and detailed models of the interface hardware and linear axis. In the emulation mode, the simulation is performed on the computer hardware that will implement the controller. In the implementation mode, the controller is deployed on the hardware system and experimental data is gathered. The physical linear axis position can be adjusted by pushing the Jog button on the main interface as shown in figure 4.2. Pushing the Jog button opens up the Jog interface as shown in figure 4.3. This interface allows the student to move the linear axis in the positive or negative direction in three increments. The target computer then acts as an interface between the hardware. This utilizes the XPC real-time operating system.

Nodel Build Run	Jog	The 201 Desire under a Davide method by Marry New Age Under ander type, Saled The reader have the reader Help The Sale a service to their a service and the Sale a service to the service and the Sale and the
ndevškole Dučke Tgan * Doceste * Ceterana Mode		Onest Sester Moto Deuter C.Donmetti and EntrymoloDevicy/RDS 12 15 bez
Sealan A	3	Searchede whe impress ed
Te-sul		View Simulink Model
forum Water Controller For an one	tes to	Tige overlager towe Click Mass onsailer "Run" to More adiatives Continue
Kulturyun Familia	3	Concerner Concerner Reference annual Discal

Figure 4.2. The RDS Phase II main interface.

Minor improvements were made to the RDS interface from Phase II such as the inclusion of an email button so that students could email the resulting plots and data sheets. A larger evaluation was conducted using more subjects for this round of the evaluation. The participants were undergraduate students from the Mechanical Engineering department who were taking the control systems class.



}

Figure 4.3. The Jog function interface.

Build Run Jog Help	Sher ACC Sher to 1 Sheri Ris simplifies directions and the directions and the	Articlesis the Secondard result to following theory of the end Articlesis for the collection pull. Collect the physics particle (Constants the collection, Phys. and Const. for the spaces, Constants the collection, Phys. and Const. for the theory of the collection of the collection of the for theory often.
Stanuards Mindael Operative	Mare Type	Analysis and Frances
	Patrimental Anodali Types Contenents	Total Total Table Constant Table Constant T
	Channelle Channelle 3.5 and	Load Model to Turges
	60 24	Similate
		STOP

Figure 4.4. The simulate function.

5. DATA COLLECTION

5.1. OVERVIEW

In this section, we discuss how participants were recruited for testing in Phase II A and in Phase II B. We then discuss the research procedure that was followed as a part of the testing of the RDS system.

5.2. PARTICIPANTS

The participants for this research were 34 (31 male and three female) students enrolled in their senior year in the "ME 279: Automatic Control of Dynamic Systems" class.

5.3. RESEARCH PROCEDURE

First, the participants were given a pre-questionnaire that contained questions on demographics, learning styles, and interest in science. This pre-questionnaire was collected from all participants one month prior to the evaluation. As a part of the evaluation, students were asked to develop their controller to use with the RDS system (this was incorporated as part of a class assignment).

In the evaluation process, each participant was given a brief run-through on the RDS system. After this run-through, the entire setup was introduced to the participant. Each subject was then given a task list that included four tasks:

1. Use the Jog function (eye tracking was used).

- 2. Simulate your controller.
- 3. Emulate your controller.
- 4. Implement your controller (eye tracking was used).

Next, every participant underwent an automatic eye-tracking calibration. After this calibration, the participant was asked to perform the tasks. The participant's eye movement data was collected using the eye tracker while they were performing the tasks. Eye fixation data and the number of fixations were obtained using the eye tracking software.

After each task, the recordings were stopped and new recordings were performed before the beginning of the next task. When all of the tasks were completed, each participant was asked fill out a questionnaire that contained questions on technology acceptance constructs, learning outcome constructs, and interest in science constructs (Appendix A). Each participant was then interviewed regarding their background with control systems, issues using the system, and recommendations for the system.

5.4. INSTRUMENT

All of the questions in the questionnaires (both pre/post) were adopted from previous literature and were 7-pointLikert scale-based questions.

The pre-questionnaire contained questions on learning style (see Appendix A). These learning style questions were adapted from "The Index of Learning Styles Questionnaire" (Felder & Soloman, 2001). The survey contains questions related to four domains: (a) visual or verbal, (b) active or reflective, (c) sensitive or intuitive, and (d) sequential or global. For this study, only the questions concerning visual and verbal learners were taken into consideration.

Felder et al. (2005) found estimates of a reliability score from 0.56 to 0.77 using the Cronbach's Alpha statistical technique. In an unpublished study, Felder and Spurlin (2005) and Livesay et al. (2002) examined the Index of Learning Styles survey responses from 584 learners at North Carolina State University and found Cronbach's alpha coefficients to be in the range of 0.55 to 0.76.

The post questionnaire (Appendix 2) contained questions from three constructs: (a) interest in science, (b) the Technology Acceptance Model (TAM), and (c) learning outcome. The questions from the TAM had questions on the perceived ease of use (Davis, 1989), the perceived usefulness (Davis, 1989), the attitude towards the system (Shirley & Todd, 1995), and the intention to use the system (Gefen et al., 2003). The questions for the learning outcome were adopted from Hall et al (2006).

For measuring attention on different parts of the interface, we created an area of interest on the part of the interface that we were interested in testing. We then calculated the total fixation duration on an area of interest. This is one of the most popular measures used by researchers to measure a user's attention towards an interface element in the area of multimedia learning with graphics (Mayer, 2010). To rephrase, this metric measures the sum of all fixation durations that are within the area of interest. The total fixation time on the relevant areas reveals the perceptual processing during learning. A study by Schmidt-Weigand et al. (2010) used fixation time on visualization and text and found that learners spent more time viewing fixations with spoken text than with written text.

Another useful byproduct of the eye tracking data is the pupil dilation measure. A study by Gerven et al. (2002) found that mean pupil dilation is useful in measuring the task evoked-pupillary response i.e. cognitive workload, especially for young adults. Various studies have shown a reliable link between cognitive processing and changes in pupil dilation (Hess & Polt, 1964; Nakayama et al., 2002; Iqbal et al., 2005).

5.5. DATA ANALYSIS

Data obtained from questionnaires was analyzed by performing t-tests, one way ANOVA, multivariate tests, correlation and regression using SPSS. The TAM model parameters were tested using regression and correlation. The perceived learning outcome was calculated by taking the average of the constructs such as learning from the lab, motivation from the lab, real world learning from the lab, and knowledge gained after doing the lab. The comparison between perceived learning outcomes for different modes was analyzed using t-tests and multivariate tests. The knowledge gained before and after using the system was compared using paired t-tests.

Attention measures were obtained from the eye tracking data. An area of interest (AOI) was created over the animation part of the interface as shown in Figure 5.1. The total fixation duration was calculated by adding all the fixation durations together inside the area of interest.

Pupil dilation was calculated individually using the pupil diameter values obtained from the eye tracker. For each trial, a baseline pupil diameter was determined by calculating the average pupil size during a period of 100 milliseconds preceding the onset of the interface.



Figure 5.1. Area of interest: animation

The pupil dilation was calculated by subtracting this baseline from the pupil diameter for each data point until the completion of the task. This difference score was converted to a percentage of the corresponding baseline value (Moresi et al., 2008).

Pupil Dilation= Pupil diameter – Baseline x 100

Pupil diameter

6. QUESTIONNAIRE DATA RESULTS

6.1. SECTION OVERVIEW

In this section, we discuss the results of the questionnaires that measure the following constructs from the research model: learning styles, interest in science, technology acceptance model and perceived learning outcome.

6.2. DEMOGRAPHICS

A total of thirty four students participated in the RDS Phase II B study. Of the thirty four students, 31 of them were male while 3 were female.

6.3. TASK PERFORMANCE

Almost all of the participants were able to complete all the tasks without the need of a prompt (table 6.1.). According to UA7, task performance measures indirectly affect the person's attitude towards the system.

1	able	6.1.	lask	performance
---	------	------	------	-------------

Task	Description	Number	of	participants
1	Use the Jog function.		33	
2	Simulate and select the controller. Generate		34	
3	Emulate and select the controller. Generate	e	34	
4	Go back and implement a preexisting	2	34	

6.4. QUANTITATIVE ANALYSIS

6.4.1. Learning styles. The chart in figure 6.1.validates previous research that most engineering students are visual, sensing, and active learners (Felder & Brent, 2005). Of the 34 students who participated in this study, 26 of them were visual learners, 20 of them were active learners, 31 were sensory learners, and 31 were sequential learners.



Learning Styles %

Figure 6.1. Learning styles.

6.4.2. Interest in science. The analysis of interest in science constructs pre (before being exposed to RDS) and post (after being exposed to RDS) showed significant improvements (see figure 6.2.). The category of career interest in science showed the maximum improvement followed by the enjoyment in science.

We also found an increase in the pre-post for General interest in science. However t-test revealed no significant changes. Paired samples t-tests were conducted on the interest in science constructs for the pre and post questionnaires. The results indicate a significant difference in career interest in science (t = -3.848, p = .001) and enjoyment in science (t = -3.198, p = .003).



Figure 6.2. Interest levels in science.

The results of the test are shown in table 6.2. below.

Table 6.	2. Paired	samples	t-test
----------	-----------	---------	--------

	Paired	Difference			Sig.			
		Std.	Std. Error	95% Confidence Interval of the Difference			16	(2- tailed
	Mean	Deviation	Mean	Lower	Upper	t	df)
Pair Pre vs. post 1 GS	19	.74	.12	45	.06	-1.49	33	.14
Pair Pre vs. post CS 2	54	.82	.14	83	25	-3.84	33	.00
Pair Pre vs. post ES 3	46	.85	.14	76	17	-3.19	33	.00

6.4.3. Perceived learning outcome. An analysis of the results from the perceived learning outcome showed the real world applicability of the RDS system to be high in comparison with lecture and notes. The results are shown in figure 6.3. below.



Figure 6.3. Comparison between learning through RDS, lecture, and notes.

In order to test the significance between the three variables i.e. learning through lectures The results show a significant multivariate effect for real world applicability and learning, notes and labs (RDS) a multivariate test was conducted. A one-way MANOVA revealed a significant multivariate main effect for real world applicability (Wilks' $\lambda =$.54, F (2, 32) = 13.648) and learning (Wilks' $\lambda = 0.815$, F (2, 32) = 3.621). The results of the multivariate tests are shown in table 6.3.

					Err	
				Hypothesi	or	
		Value	F	s df	df	Sig.
Learning	Pillai's Trace	0.18	3.62a	2	32	0.03
	Wilks' Lambda	0.81	3.62a	2	32	0.03
	Hotelling's Trace	0.22	3.62a	2	32	0.03
	Roy's Largest Root	0.22	3.62a	2	32	0.03
Motivation	Pillai's Trace	0.03	.59a	2	32	0.55
	Wilks' Lambda	0.96	.59a	2	32	0.55
	Hotelling's Trace	0.03	.59a	2	32	0.55
	Roy's Largest Root	0.03	.59a	2	32	0.55
Real World						
Applicability (RWE)	Pillai's Trace	0.46	13.64a	2	32	0
	Wilks' Lambda	0.54	13.64a	2	32	0
	Hotelling's Trace	0.85	13.64a	2	32	0
	Roy's Largest Root	0.85	13.64a	2	32	0
a. Exact statistic						
b. Design: Intercept						

Table 6.3. Multivariate test results

A one way ANOVA was conducted for learning, motivation and real world applicability constructs of perceived learning outcome for comparing the perceived learning outcome between lecture, lab and notes. The results showed that there was a significant difference (p=0.001) in the means of real world applicability between lecture, labs and notes. A Turkey post-hoc test revealed that the perceived real world applicability through labs was significantly different from notes (p=0.000).There was no significant difference in the perceived learning and motivation between the three modes of instruction i.e labs, lectures and notes. The results of one way ANOVA are shown in table 6.4. below.

		Sum of		Mean		
		Squares	Df	Square	F	Sig.
Motivation	Between Groups	.54	2	.27	.17	.83
	Within Groups	152.94	99	1.54		
	Total	153.49	101			
RealWorld	Between Groups	20.25	2	10.12	7.72	.00
	Within Groups	129.82	99	1.31		
	Total	150.07	101			
Learning	Between Groups	10.29	2	5.14	2.38	.09
	Within Groups	213.91	99	2.16		
	Total	224.2	101			

Table 6.4. One way ANOVA results for perceived learning outcome

6.4.4. Knowledge gained. The students also felt that they gained an increase in their perceived knowledge after using RDS, which establishes the fact that RDS could be used as part of the course-curriculum and which proves its role as a learning tool. The chart in figure 6.4.shows the knowledge gained by the students before and after using RDS. However it should be noted that we measured the perceived knowledge from the perceived learning outcome (Appendix B).

In order to test the significance between both the constructs i.e. perceived knowledge prior to using RDS system and perceived knowledge after using the system we conducted a paired samples t-test.



Figure 6.4. Knowledge gained before and after using RDS.

Paired t-test conducted on the knowledge before and after using RDS revealed significant increase. The results of the paired t-test showed a significant difference in the perceived knowledge gained (t= -6.076, p = .000) and are shown in table 6.5.

	Paired I	Differences						
		Std.	Std. Error	95% Co Interval Differenc	onfidence of the e	+	df	Sig. (2- tailed)
	Mean	Deviation	Mean	Lower	Upper		u	(ancu)
PairKnowledge Gained 1 Before Using RDS – Knowledge Gained After Using RDS	-1.26	1.21	.2	-1.68	84	-6.07	33	.00

Table 6.5. Paired samples t-test for perceived knowledge gained

6.4.5. Technology acceptance constructs (TAM). Results from the Technology acceptance constructs revealed that students rated the RDS system as easy to use, useful in their jobs and had favorable attitude towards the system. The results also show relatively low intention to use RDS. One of the plausible reasons for this can be attributed to the fact that all the participants were undergraduate students and most of them did not expect to use RDS in classes other than in the senior year class (ME 279). Figure 6.5. shows the values of various TAM constructs.

Regression analysis was conducted between attitude as dependent variable and perceived usefulness and perceived ease of use and independent variables. The model was found to be significant ($R^2=0.812$) and accounted for 80% variance in the attitude towards the system.



Figure 6.5. Technology acceptance constructs.

Perceived usefulness significantly predicted the attitude towards the system (p=0.000, β =0.105). The results of the regression analysis are presented in table 6.6 and 6.7 below.

				Std. Error	Change Statistics				
		R	Adjusted	of the	R Square	F			Sig. F
Model	R	Square	R Square	Estimate	Change	Change	df1	df2	Change
1	.90 ^a	.81	.80	.45	.81	67.14	2	31	.00

Table 6.6. Regression model summary

a. Predictors: (Constant), PU, PEOU

Table 6.7. Coefficients of regression

		Unstandardized Coefficients		Standardized Coefficients		
Model		В	Std. Error	Beta	t	Sig.
1	(Constant)	.03	.48	a	.07	.94
	PEOU	.13	.11	.11	1.17	.24
	PU	.87	.10	.82	8.33	.00

In order to test the relationship between the various constructs of the technology acceptance model & the intention to use a correlation test was conducted between attitude towards the system, intention to use the system and perceived learning outcome. The results revealed that the correlation was significant between attitude, intention to use and the perceived learning outcome. This shows that the constructs are highly related and thus proves strong relationship between all the constructs used in the model.

The results of the correlation are presented in table 6.8. below.

		Attitude	Intention	LO
Attitude	Pearson	1	.62**	.46**
	Correlation			
	Sig. (2-tailed)		.00	.00
	Ν	34	34	34
Intention	Pearson	.62**	1	.40*
	Correlation			
	Sig. (2-tailed)	.00		.01
	Ν	34	34	34
LO	Pearson	.46**	.40*	1
	Correlation			
	Sig. (2-tailed)	.00	.01	
	Ν	34	34	34

Table 6.8. Results of correlation

**. Correlation is significant at the 0.01 level (2-tailed).

*. Correlation is significant at the 0.05 level (2-tailed).

7. EYE TRACKING RESULTS

7.1. SECTION OVERVIEW

In this section we discuss the qualitative findings from the eye tracking methodology and report usability issues with the interface. We utilize the quantitative data from the eye tracking to study the impact of the use of the animation on the attention of two learning styles of students: visual and verbal. Pupil dilation measures from the eye tracking data are used to study if there is any significant difference in the cognitive workload between the two learning groups. For this part of the results, we used the combined data from two evaluations to increase the sample size and to improve the validity of the results. The same interface was used in both the evaluations.

7.2. DEMOGRAPHICS

A total of 50 students participated in the evaluation. Of these, 39 were visual learners and the remaining 11 were verbal learners.

7.3. RESULTS

7.3.1. Overview. The eye tracking results were used to triangulate the results from the survey data and to collect objective eye movement data on the interface. From a qualitative point of view, the eye tracking data provided us with two important visualizations: (a) gaze plots and (b) heat maps. Gaze plots display a static view of the gaze data for each image of the stimuli. It helps in visualizing the scan path of the

participant over the interface. Each fixation is illustrated with a dot and the radius of the dot represents the length of the fixation. Each dot is embedded with a number that indicates the order of the fixation.

Another powerful visualization is the heat map. The heat map allows researchers to visualize the gaze behavior of an entire group. A heat map is composed of a background image (stimulus) on which a heat map layer is superimposed. The heat map layer is obtained by combining the gaze data from multiple participants. To differentiate between looking behavior, heat maps uses the temperature analogy to reflect higher fixations versus lower fixations. In the heat maps generated from our data, we used the color red to indicate higher fixations and the color green to indicate lower fixations.

In this section, we divide the results from the eye tracking data into two parts: (a) qualitative where we discuss the results from the gaze plots and heat maps and (b) quantitative where we discuss the attention and cognitive workload measure with respect to the survey data.

7.3.2. Qualitative analysis. In order to help us easily deduce the results, we reduced the number of fixation data points by shrinking the timeline to the first five seconds for a few of the results. From Figure 7.1., we noticed that the user immediately noticed the Jog button as part of the first task in which the users had to use the Jog function. It can be seen that, based on the order of fixations 1, 2 and 3, the user was able to easily identify the Jog function button.

In the Figure 7.2., we see that the user did not read the pop-up message that appears when a user is saving the data. The pop-up message box has important information on how to understand the saved data when it is opened elsewhere. We found that the users generally dismissed the pop-up message box without reading it. Generally, in gaze plots on pop-up message boxes, researchers see fixations on the message which indicates the user has read the message.

1	Line	ar Axis Rapid Deve	kopensent S	yistem - Builde		3-
	Mouri Fuld	me Di	V Selat		ACCUSE.	+
	The local data		-			
	-					
	1000		-	12 -	Noi	
	-		Louis	-		
		and a start of the	(int)	-		
	Reporter Stores	utions Advintes				

Figure 7.1. User notices the Jog button.



Figure 7.2. Gaze plot shows user did not notice pop up message.

However, in this case, we did not notice any fixations on the message, which suggests that the users did not read it. The figure 7.3. shows the heat map related to the pop- up message.



Figure 7.3. Heat map show users did not notice pop up message.

In Figure 7.4., we see from the gaze plot that the users did not read the pop-up dialog box that suggests that the user verify whether he/she has checked the help page before inserting the controller. Most of the users instantly dismissed this message as soon as they encountered it as shown in figure 7.5. Therefore, the interface should minimize the number of pop-up message boxes and use them only for displaying important messages.

Figure 7.6. shows the gaze plot of a participant on encountering an error message, which popped up when XPCTargetTM, specialized software that enables students to execute Simulink models, was not connected to the target CNC system.

Lines Model Burd	Run Jog	velopement S	rstern - Build	
yah galaan bah juli kunstapi "2006 	- 90-	ander og geden ander den ander og	n terret	ware were free
			-	fict. Reactly
	va	11	Ξ	

Figure 7.4. Gaze plot shows user did not notice pop up dialog box.



Figure 7.5. Heat map shows users did not notice pop up dialog box.

The gaze plot as seen in figure 7.6. indicates that the user is frustrated, which can be noticed from the rapid saccade movement. Previous research in the area eye tracking with respect to interface usability attributed rapid saccade movement over an interface to difficulty in understanding how to operate the interface (Goldberg & Kotval, 1999).



Figure 7.6. Rapid saccade movements.

An analysis of the heat maps suggests that all the important areas of the RDS interface received copious attention. As can be seen in figure 7.7., the red indicates that higher attention was paid towards the animation part of the interface. Since the buttons were designed to be highly usable with a large font size and a large button size, the users were able to notice the buttons easily. In this heat map, we notice that the area around the Jog function button has a higher focus as this was first task where users had to use the Jog function. On the right side is the operation panel to control the Jog function. This also

received copious attention from the users as it is an important part of the task and the interface.



Figure 7.7. Heat map of Jog function.



Figure 7.8. Heat map of help menu.

Figure 7.8. shows the heat map of the help menu. An analysis of the heat map suggests that the users were able to read the help menu easily. However, many users did not scroll to the bottom part of the help menu where pictures provided additional guidance.

7.3.3. Quantitative analysis. For this part of the analysis, we used the quantitative measures from the eye tracking data. We used the total fixation duration to measure the attention towards the animation part of the interface. The pupil dilation measure was used to measure the cognitive workload on the interface. We then used these measures in combination with the survey data to study the impact of learning styles on the perceived learning outcome, the perceived ease of use, attention, and cognitive workload. We use the methods outlined in data analysis (section 5.4) for extracting attention and pupil dilation measures.

7.3.3.1. Effect of learning style on attention towards animation. The moderating role of learning styles on the student's attention towards the animation part of the interface was measured using one way ANOVA. This analysis revealed that the attention towards the interface differed significantly between both the groups [F (1, 44) = 6.929, p = 0.017]. The results are shown in table 7.1. below.

	Sum of		Mean		
	Squares	Df	Square	F	Sig.
Between	.02	1	.02	6.12	.01
Groups					
Within Groups	.16	44	.00		
Total	.19	45			

Table 7.1. ANOVA for attention towards animation

7.3.3.2. Effect of learning style on cognitive workload. A one-way-betweensubjects ANOVA was conducted to compare the difference in the cognitive workload between the visual and verbal learning groups. A significant difference was observed between both the groups at the p<.05 level for the three conditions [F (1, 44) = 31.427, p = 0.00]. The results are shown in table 7.2. below.

Sum of Mean Squares Df F Sig. Square Between .008 1 .008 31.42 .00 Groups Within Groups .011 44 .00 Total .019 45

Table 7.2. ANOVA for cognitive workload

7.3.3.3. Effect of learning style on perceived ease of use. A one-way betweensubjects ANOVA was conducted to compare the difference in the perceived ease of use (PEOU) between the visual and verbal learning groups. The one-way ANOVA revealed significant differences both the groups. Result showed difference between both the groups i.e. visual and verbal at the p<.05 level for the two conditions [F (1, 44) = 24.040, p = 0.00].

The results are shown in table 7.3. below.

	Sum of Squares	Df	Mean Square	F	Sig.
Between Groups	15.67	1	15.67	24.04	.00
Within Groups Total	28.68 44.35	44 45	.65		

Table 7.3. ANOVA for perceived ease of use

7.3.3.4. Differences in perceived learning outcome. A one-way ANOVA between subjects was conducted to compare the difference in the perceived learning outcome between the visual and verbal learning groups. There was a significant difference between the perceived learning outcome for both the groups at the p<.05 level for the three conditions [F (1, 44) = 10.352, p = 0.02]. The results are shown in table 7.4. below.

	Sum of		Mean		
	Squares	Df	Square	F	Sig.
Between	7.48	1	7.48	10.35	.00
Groups					
Within Groups	31.79	44	.72		
Total	39.28	45			

Table 7.4. ANOVA for perceived learning outcome

8. DISCUSSION AND IMPLICATIONS

8.1. DISCUSSION

In this thesis, we conducted a usability evaluation of a computer-based learning system that was designed to impart practical knowledge and serve as a complementary instructional laboratory to the theoretical class. We utilized eye tracking in conjunction with surveys in evaluating the Rapid Development System, a computer-based learning system for teaching control design/insertion in the mechanical engineering curriculum. The test validated the usefulness of the eye tracking technology in providing additional insights into users' cognitive aspects while interacting with the learning system. Overall, the results from the research suggest that the RDS system was well received by the participants.

The results from the study related to the learning styles validated previous research findings that most engineering students are visual learners (i.e. they prefer learning through a visual medium), which bodes well for the highly visual RDS interface. We found that the students had a higher interest in taking up careers in science after using the system. Students reported an increase in enjoyment in science after using the system. Result also showed high real world applicability of RDS system in comparison to lectures and notes. Students also reported an increase in their perceived knowledge after using the system.

The eye tracking results validated the results from the survey analysis. The gaze plots and the heat maps indicated that the participants were able to identify important areas of the interface, such as the tip box and help button, which were newly developed.

However, we noticed a few minor usability issues with the interface for which we provide recommendations. Implementing these changes should be helpful in the further refinement and improvement of the RDS interface.

The eye tracking data indicated that users did not read the dialog message on the pop-up box. This was observed in the eye tracking data of multiple participants. Our suggestion to the design team would be to increase the size of the dialog box and to increase the font size of the message. It is well-known in the area of user experience that increasing the size of the interface element will increase the user's attention towards it (Fitts, 1954). Another tip for the developers is to include with the legend information in the data file, so the students can easily understand the data output.

The eye tracking results also indicated that users did not read the message that pops up when they are inserting the controller. Our suggestion is to remove this pop-up box.

Again, results from the eye tracking data indicated that one user was frustrated when the error "Connecting XPC to target" popped up on the user's screen. This was observed from the gaze plot of a user where the user's rapid saccade movement was observed. Researchers have found that saccadic eye movements precede attention and that a close relationship exists between them when processing complex information tasks (Hoffman & Subramaniam, 1995; Stelmach & Herdman, 1997). Research in usability using the eye tracking method attributes rapid saccade movement over an interface to difficulty in understanding (Goldberg & Kotval, 1999).

The development team should ensure that the error message "Connecting XPC to target" does not pop up when the user is performing tasks since it can increase the
cognitive workload on the user. We suggest that the developers include a button to connect the XPC to the target directly or that the user should be transferred directly to the help page that provides instructions on how to reconnect the system once the user clicks on the dialog button.

Before students begin using the RDS system, a brief mini-video should be shown to explain the RDS system and to provide an overview on how it can be used, its capabilities, and its rich feature set. This video would help to reduce the initial cognitive workload of students using the system.

The quantitative data from the eye tracker provided us with metrics that helped us in studying the differences between both the learning groups with respect to attention and cognitive workload. We also looked at other constructs such as the learning outcome and the perceived ease of use. The results indicated that the learning style of the student played a significant role in determining their attention towards the interface and their cognitive workload. The visual students had a lower cognitive workload than the verbal students did. The results from the study confirmed objectively the previous finding that visual learners prefer visual medium (Felder &Silverman, 1988). However, more research is required to study the effect of learning style on attention, cognitive workload, and other constructs.

8.2. IMPLICATIONS

The results from this research prove once again that usability testing with the end users is very important while designing user interfaces. Usability testing helps to identify problems with the interface and provides valuable cues to make the interface more usable. The results from the learning styles of the students reaffirm previous research that most of the engineering students are visual learners. Hence, educational interfaces for engineering students should be designed with consideration of the learning styles of students.

The results from our study also confirm previous findings that computer-based learning environments help increase the amount of learning when they are used as part of the educational curriculum. We also found that students reported having a higher interest in science after using the system. Both of these results suggest that additional courses and disciplines in engineering should encourage the use of computer-based learning environments to inculcate the practical training of theoretical concepts.

The initial results from our study suggest that the differences between visual and verbal learners have an impact on how these groups perceive the attention paid towards the interface, the cognitive workload, and the perceived learning. These results suggest that designers should carefully cater to each of these groups by designing various features into computer interfaces that assist the learning styles of both these groups. However, additional research should be conducted with a larger sample size to confirm these findings. Also, additional research should be conducted with other categories of learners such as sensory versus intuitive learners and active versus reflective learners.

This study also helps prove the reliability of the research model that was used to guide the testing. We believe that this model can be valuable for guiding the evaluation of other educational learning systems.

9. CONCLUSION

The results of this study provide valuable suggestions to the developers and designers of computer-based learning environments to include features that would make interfaces more usable. Regardless of the learners' educational background, instructors have an enormous task meeting individual learning styles while teaching. Using CBLEs as part of a course to increase student interest in the course and to provide them with real world learning will help instructors to accomplish this task successfully.

Our research also reaffirms the importance of including usability testing with target users as a hallmark of designing user-centered design philosophy. The results from the eye tracking reports validate the significance of using this methodology in evaluating interfaces.

The results from the study also validate the research model that was used to guide the evaluation. Further collaboration with other universities and researchers should be sought to help in further refining the model to allow educational researchers to use our model in guiding their evaluation of learning systems.

The results from our study confirm that the use of the RDS system helped to improve students' knowledge and resulted in an increase in real world learning. Our results also confirm previous findings that computer-based learning systems should be used as a part of the learning curriculum.

We also found few minor usability issues with the RDS system, which should be eliminated to further increase the usability of the system. By increasing the usability of the system, the cognitive workload can be decreased which would enable students to work distraction-free and, thus, would maximize learning. We also provide recommendations on new features, which, if followed, could help increase the appeal of the system.

The results of the eye tracking quantitative data yielded statistically significant differences between the visual and verbal learners with respect to their cognitive workloads and attention levels. These results suggest that interface designers should use various strategies to reduce the cognitive workload and to increase a user's attention to relevant areas in order to maximize learning.

However, it should be noted that this research had a few limitations. The sample size for the verbal group was very small in comparison to the visual group. The use of physiological measures such as pupil dilation can be contaminated by subjective and environmental factors. Hence, additional detailed research should be conducted using large sample sizes and to further control the external factors that can affect subjective measures.

Further research in this area should look at how other dimensions of the learning styles, such as active versus reflective learning and sequential versus global learning, to gain a better understanding of how these characteristics affect the attention and the cognitive workloads of students.

APPENDIX A. PRE QUESTIONNAIRE Please circle the answer that applies best to you.

- 1. When I have to work on a group project, I first want to
- (a) Have "group brainstorming" where everyone contributes ideas.
- (b) Brainstorm individually and then come together as a group to compare ideas.
- 2. In reading nonfiction, I prefer
- (a) Something that teaches me new facts or tells me how to do something.
- (b) Something that gives me new ideas to think about.
- 3. When I get directions to a new place, I prefer
- (a) A map.
- (b) Written instructions.
- 4. When writing a paper, I am more likely to
- (a) Work on (think about or write) the beginning of the paper and progress forward.
- (b) Work on (think about or write) different parts of the paper and then order them.
- 5. When I am learning something new, it helps me to
- (a) Talk about it.
- (b) Think about it.
- 6. I am more likely to be considered
- (a) Careful about the details of my work.

- (b) Creative about how to do my work.
- 7. When I see a diagram or sketch in class, I am most likely to remember
- (a) The picture.
- (b) What the instructor said about it.
- 8. It is more important to me that an instructor
- (a) Lay out the material in clear sequential steps.
- (b) Give me an overall picture and relate the material to other subjects.
- 9. In a study group working on difficult material, I am more likely to
- (a) Jump in and contribute ideas.
- (b) Sit back and listen.
- 10. I prefer courses that emphasize
- (a) Concrete material (facts, data).
- (b) Abstract material (concepts, theories).
- 11. When I meet people at a party, I am more likely to remember
- (a) What they looked like.
- (b) What they said about themselves.
- 12. When I solve math problems

- (a) I usually work my way to the solutions one step at a time.
- (b) I often just see the solutions but then have to struggle to figureout the steps to get to them.

APPENDIX B. POST QUESTIONNAIRE For the following questions, please indicate the extent to which you agree or disagree with the following statements.

Scale – (Strongly disagree (1) – Strongly agree (7))

- 1. Using Rapid Development System enables me to accomplish tasks more quickly.
- 2. Learning to operate Rapid Development System is easy for me.
- 3. I find it easy to get Rapid Development System to do what I want it to do.
- 4. Using the Rapid Development System is a good idea.
- 5. Using Rapid Development System improves my learning outcome.
- 6. My interaction with Rapid Development System is clear and understandable.
- 7. I predict to use the Rapid Development System in the next six months.
- 8. Using Rapid development system would make it easier to understand control design/insertion.
- 9. I find Rapid Development System to be flexible to interact with.
- 10. I like the idea of using the Rapid Development System.
- 11. I find Rapid Development System useful in learning control design/insertion/insertion.
- 12. It is easy for me to become skillful at using Rapid Development System.
- 13. I intend to use the Rapid Development System in the next six months.
- 14. I find Rapid Development System easy to use.
- 15. Using the Rapid Development System would be pleasant.
- 16. I plan to use the Rapid Development System in the next six months.
- 17. I learned a great deal of information about control design/insertion from this week's lab.

- I learned a great deal of information about control design/insertion from class lectures.
- 19. I learned a great deal of information about control design/insertion from the class text.
- 20. I found this week's lab on control design/insertion to be very motivational.
- 21. I found the class lectures over control design/insertion to be very motivational.
- 22. I found the class textbook's coverage of control design/insertion to be very motivational.
- 23. This week's lab activity over control design/insertion was applicable to "real world" engineering.
- 24. The class lecture over control design/insertion was applicable to "real world" engineering.
- 25. The text book coverage of control design/insertion was applicable to "real world" engineering.
- 26. Before the lab activity that covered control design/insertion, I knew a great deal about the subject.
- 27. After the lab activity that covered control design/insertion, I knew a great deal about the subject.
- 28. I would like to belong to a science club.
- 29. I would dislike being a scientist.
- 30. Science lessons are fun.
- 31. I get bored watching science programs on TV.
- 32. I would like to work with people who make discoveries in science.
- 33. I dislike science lessons.

- 34. I would like to be given a science book or a piece of scientific equipment as a present.
- 35. I would dislike a job in a science laboratory.
- 36. School should have more science lessons each week.
- 37. I dislike reading books about science in my leisure time.
- 38. Science lessons bore me.
- 39. Working in a science laboratory would be an interesting way to earn a living.
- 40. Science is one of the most interesting school subjects.

REFERENCES

Ajzen, I. (1991). The theory of Planned Behavior, *Organizational Behavior and Human Decision Processes*, 50 (2), 179–211.

Anderson, J. R. (2004). Cognitive psychology and its implications (6th ed.). Worth Publishers.

Ardito, C., Costabile, M.F., De Marsico, M., Lanzilotti, R., Levialdi, S., Roselli, T., and Rossano, V. (2006). An Approach to Usability Evaluation of e-Learning Applications, Universal *Access in the Information Society International Journal*, 4 (3), 270–283.

Azevedo, R., Guthrie, J, T., Seibert, D. (2004). The Role Of Self-Regulated Learning In Fostering Students' Conceptual Understanding Of Complex Systems With Hypermedia, *Journal of Educational Computing Research*, 30(1), 87-111.

Ayres, P., Sweller, J. (2005). The split-attention principle in multimedia learning. In R. E. Mayer (Ed.), The Cambridge handbook of multimedia learning (pp. 134–146). New York: Cambridge University Press.

Beatty, J. Lucero-Wagoner B. (2000). The pupillary system. In J. Caccioppo, L.G. Tassinary, & G. Berntson (Eds.) The Handbook of Psychophysiology, Hillsdale, NJ: Cambridge University Press.

Bradshaw, J. L. (1968a) Load and papillary changes in continuous processing tasks, *British journal of psychology*, 59, 265-271.

Bradshaw, J. L. (1968b) Pupil size and problem solving, *Quarterly Journal of Experimental Psychology*, 20, 116-122.

Benbunan-Fich, R. (2002). Improving education and training with IT, *June Communications of the ACM*, *45*(6), 100-114.

Bevan, N. (2001). International standards for HCI and usability, *International Journal of Human Computer Studies*, 55(4), 533-552.

Bovy, R.C. (1981). Successful instructional methods: a cognitive information processing approach. *ECTJ*, 29(4), 203–217.

Bransford, J. D., Vye, N., Kinzer, C., Risko, V. (1990). Teaching thinking and content knowledge: Toward an integrated approach. In B. F. Jones & L. Idol (Eds.), *Dimension sof thinking and cognitive instruction: Implications for educational reform* (Vol. 1, pp.381-413). Hillsdale, NJ: Lawrence Erlbaum Associates.

Cairncross, S., Mannion, M. (2001). Interactive multimedia and learning: Realizing the benefits. *Innovations in Education & Teaching International*, *38*(2), 156-164.

Carlson, L. E., Sullivan, J. F. (1999). Hands-on engineering: Learning by doing in the integrated teaching and learning program, *International Journal. Engineering Education*, *15*(1), 20–31.

Carver, C.A, Howard, A. R., Lane, W. D. (1999). Enhancing student learning through hypermedia courseware and incorporation of student learning style, *IEEE Transactions on Education*, *42*(1), 33-38.

Chen, J., The effects of education compatibility and technological expectancy on elearning acceptance, *Computers & Education*, 57, 1501-1511.

Colman, F. D., Paivio, A. (1969). Pupillary response and galvanic skin response during an imagery task, *Psychometic Science*, 16, 296-297.

Clough, M. P. (2002). Using the laboratory to enhance student learning. In R. W. Bybee (Ed.), *Learning Science and the Science of Learning*, pp. 85–97.

Corno, L., Snow, R.E. (1986). Adapting teaching to individual differences among learners. In M. C. Wittrock (Ed.), *Handbook of research on teaching*(3rd ed.). New York, N.Y.: Macmillan and Co.

Corter, J. E., Nickerson, J. V., Esche, S. K., Chassapis, C., Im, S., Ma, J. (2007). Constructing reality: A study of remote, hands-on, and simulated laboratories, *ACM Transactions on Human Computer Interaction.*, *14*(2), Article 7.

Cuomo, D. L. and Bowen, C. D. (1994). Understanding usability issues addressed by three user-system interface evaluation techniques, *Interacting with Computers*, 6(1), 86–108.

Danao, G. (2010). The SMU eLearning System: A Students Usability Assessment, *SCSIT Research Journal*, 1(1), 63-70.

Davis, F. (1989).Perceived usefulness, perceived ease of use, and user acceptance of information technology, *MIS Quarterly*, 30(3), 323-340.

de Koning, B. B., Tabbers, H. K., Remy, J., Rikers, M. J. P., &Paas, F. (2010). Attention guidance in learning from a complex animation: Seeing is understanding?, *Learning and Instruction*, *20*(2), 111-122.

Desimone, R., Duncan, J. (1995).Neural mechanisms of selective visual attention, *Annual Review Neuroscience*, *18*, 193-222.

Desurvire, H. W. (1994). "Faster, cheaper!! Are usability inspection methods as effective as empirical testing?," In *Usability inspection methods*.

Dutt, A., Johnson, H., Johnson, P. (1994). Evaluating evaluation methods, *People and Computers IX*. BCS HCI conference, Cambridge University Press.

Edelson, D. C., Gordin, D. N., & Pea, R. D. (1999). Addressing the challenges of inquirybased learning through technology and curriculum design. *Journal of Learning Science*, 8(3&4), 391–450.

Eder, W. E., Hubka, V. (2005).Curriculum, pedagogies and didactics for design education, *Journal for Engineering Design*, *16*(1), 45-61.

Evans, C. & Edwards, M. (1999).*Navigational Interface Design for Multimedia Courseware*. Journal of Educational Multimedia and Hypermedia, 8 (2), 151-174.

Evans, C., Gibbons, N., Shah, K., Griffin, D.K. (2004). Virtual learning in the biological sciences – pitfalls of simply "putting notes on the web". *Computers & Education*, 43(1-2), 49-61.

Feisel, L.D., Rosa, A. J. (2005). The role of laboratory in undergraduate engineering education, *Journal of Engineering Education*, 4(3), 30-42.

Felder, R. M. (1996). Matters of style, *AEEE Prism*, 6(4), 18-23.

Felder, R.M., Brent, R. (2005). Understanding student differences, *Journal of Engineering, Education*, 94(1), 57–72.

Felder, R.M., & Silverman, L.K. (1988). Learning and teaching styles in engineering education, *Journal of Engineering Education*, 78(7), 674-681.

Felder, R.M., Soloman, B.A. (2001). Index of learning styles questionnaire. Retrieved from http://www2.ncsu.edu/unity/lockers/users/f/felder/public/ILSdir/ILS-a.htm

Felder, R.M., J. Spurlin. (2005). Reliability and Validity of the Index of Learning Styles: a Meta-analysis, International Journal of Engineering Education, 21(1), 103–112.

Felder, R, M., Litzinger, T, A., Lee, S, H., Wise, J, C. (2005). A Study of the Reliability and Validity of the Felder-Soloman Index of Learning Styles, *Proceedings of the 2005 American Society for Engineering Education Annual Conference & Exposition.*

Finkelstein, N., Adams, W., Keller, C., Kohl, P., Perkins, K., Podolefsky, N., LeMaster, R. (2005). When learning about the real world is better done virtually: A study of substituting computer simulations for laboratory equipment. *Physical Review Special Topics -Physics Education Research*, *1*(1), 101-130.

Fishbein, M.A., Ajzen, I. (1975). Belief, Attitude, Intention and Behavior: An Introduction to Theory and Research. Addison-Wesley, Reading, MA.

Fraser, B. J. (1981). Test of science related skills, *Australian Council for Educational Research*. Hawthorn, Victoria: The Australian Council for Educational Research Limited.

Ganatos, P., Liaw, B. (1995).Computer-animated teaching software for engineering dynamics and mechanical vibration. In *Frontiers in Education Conference Proceedings*.

Gefen, D., Karahanna, E., Straub, D. W. (2003). Trust and TAM in online shopping: An integrated model.*MIS Quarterly*, *27*(1), 51-90.

Graf, S., Viola, S. R., Leo, T., Kinshuk, K. (2007). In-depth analysis of the Felder-Silverman learning style dimensions, *Journal of Research on Technology in Education*, *40*(1), 67-82.

Granić, A. (2008). Experience with usability evaluation of e-learning systems, Universal Access in the Information Society, 7(4), 209–22.

Graesser, A.C., ChiGraesser, A.C., P. Chipman, B. C. Haynes, A, Olney. (2005). AutoTutor: An intelligent tutoring system with mixed-initiative dialogue, *IEEE Transactions in Education*, 48, 612-618.

Hall, R.H., Philpot, T., Hubing, N. (2006).Comprehensive assessment of a software development project for engineering instruction, *Journal of Learning, Technology, and Assessment*, *15*(5), 4–42.

Hannafin, M. J. (1992). Emerging technologies, ISD, and learning environments: Critical perspectives, *Educational Technology Research and Development*, 40(1), 49–63.

Hannafin, M.J., Land, S. (1997). The foundations and assumptions of technologyenhanced, student-centered learning environments, *Instructional Science*, 25, 167-202.

Hartson, H.R., Andre, T.S., and Williges, R.C. (2003).Criteria for evaluating usability evaluation methods, *Human-Computer interaction*, 15(1), 145-181.

Hayes, J., Allison, C. W. (1994). Cognitive styles and its implication for management practice, *British Journal of Managements*, *5*(1), 53-71.

Horton. W. (2000). Designing Web-Based Training, John Wiley, New Jersey. Duchowski, A. T. (2003). *Eye Tracking Methodology: Theory and Practice*. London: Springer-Verlag.

Hess, E. H. (1965). Attitude and pupil size, Scientific American, 212, 46-54.

Hess, E. H., Polt, J. M. (1964) Pupil size in relation to mental activity during simple problem-solving, Science, 143, 1190-1192.

Fitts, P. M. (1954). The information capacity of the human motor system in controlling the amplitude of movement, *Journal of Experimental Psychology*, *47*, 381–391.

Goldberg, J. H., Kotval, X. P. (1998). Eye movement-based evaluation of the computer interface. In: S. K. Kumar (ed.), *Advances in Occupational Ergonomics and Safety*. Amsterdam: ISO Press. 529-532.

Goldberg, J. H., Stimson, M. J., Lewenstein, M. Scott, N., Wichansky, A. M. (2002). Eye tracking in web search tasks: Design implications. In: *Proceedings of the Eye Tracking Research & Applications Symposium 2002* (pp. 51–58). New York.

Hammoud, R. (2008). Passive Eye Monitoring: Algorithms, *Applications and Experiments*. Springer-Verlag, 100-150.

Hannus, M., Hyona, J. (1999).Utilization of illustrations during learning of science textbookpassages among low- and high-ability children, *Contemporary Educational Psychology*, 24, 95–123.

Hegarty, M., Just, M. A. (1993). Constructing mental models of machines from text and diagrams, *Journal of Memory and Language*, 32, 717–742.

Holsanova, J., Holmberg, N., Holmqvist, K. (2009).Reading information graphics: The role of spatial contiguity and dual attentional guidance, *Applied Cognitive Psychology*.

Jacob, R. J. K., Karn, K. S. (2003). Eye tracking in human-computer interaction and usability research: Ready to deliver the promises. In J. Hyona, R. Radach, and H. Deubel (eds.), *The mind's eye: Cognitive and applied aspects of eye movements.* Elsevier, London, UK, 573-605.

John, M., Kobus, D., & Morrison, J. (1905). *Augmented cognition technical integration experiment (TIE)* (Technical report).

Just, M. A., & Carpenter, P. A. (1976). Eye fixations and cognitive processes, *Cognitive Psychology*, *8*, 441-480.

Honey, P., Mumford, A. (1982). *The Manual of Learning Styles*. Maidenhead, UK, Peter Honey Publications.

Libby, W. L., Jr., Lacey, B. C., Lacey, J. I. (1973). Pupillary and cardiac activity during visual attention, *Pscychophysiology*, 3, 229-237.

Mayer, R. E. (2005). Cognitive theory of multimedia learning. In R. E. Mayer (Ed.), *Cambridge handbook of multimedia learning* (pp. 31–48). New York: Cambridge University Press.

Kahneman, D., Beatty, J. (1966) Pupil diameter and load on memory, *Science*,54, 1583-1585.

Kahneman, D., Beatty, J. (1967). Pupillary response in a pitch-discrimination task. *Perception and Psychophysics*, 2, 101-105.

Kahneman, D., Peavler, W. S. (1969) Incentive effects and pupillary changes in association learning, *Journal of Experimental Psychology*, 79, 312-318.

Kahneman, D. Tursky, B., Shapiro, D., Crider, A. (1969). Pupillary, heart rate and skin resistance changes during mental task. Journal of Experimental Psychology, 79, 164-167.

Kahneman, D. (1973). Attention and Effort. Prentice-Hall, 19-20.

Keefe, J. W. (1979). Learning style: An overview. In *NASSP's Student learning styles: Diagnosing and proscribing programs* (pp. 1-17). Reston, VA: National Association of Secondary School Principals.

Kheir, N. A., Astrom, K. J., Auslander, D., Cheok, K. C., Franklin, G. F., Masten, M., Rabins, M. (1996). Control systems engineering education. *Automatica*, *32*, 147-166.

Klassen, J., Vogel, D. R., Moody, E. (2001) Interactive learning: design and evaluation. *Proceedings of the 34th Hawaii International Conference on System Sciences*, January 03-06, Maui, Hawaii.

Knight, D. W., Carlson, L. E., & Sullivan, J. (2007). Improving engineering student retention through hands-on, team based first-year design projects. In *Proceedings of the International Conference on Research in Engineering Education*, Honolulu, HI.

Kolb, David (1984). *Experiential learning: Experience as the source of learning and development*. Englewood Cliffs, NJ: Prentice-Hall.

Kopf, D. (2007). e-Learning market to hit \$52.6B by 2010. The Journal. Retrieved April 2, 2009, from <u>http://www.thejournal.com/articles/21046/</u>.

Kozma, B. R. (1991). Learning with media. *Media Review of Educational Research*, 61(2), 179-211.

Kramer-Koehler, P., Tooney, N. M., & Beke, D. P. (1995). The use of learning style innovations to improve retention. In *Frontiers in Education Conference*.

Kriz, S., Hegarty, M. (2007). Top down and bottom up influences on learning from animations. *International Journal of Human-Computer Studies*, 65,911–930.

Kuljis, J., Liu, F. (2005). A comparison of learning style theories on the suitability for elearning. In M. H. Hamza (Ed.), Proceedings of the IASTED Conference on Web-Technologies, Applications, and Services. ACTA Press.191-197.

Kyun, L, B., Paul, L., Graff, L, W., Sheafer, V. (2009). Understanding Cognitive Style and Learning Style and Their Application to Teaching and Learning in Engineering, American Society for Engineering Education, 11(2), 85-117. Lave, J., Wenger, E. (1991). *Situated learning: Legitimate peripheral participation*. Cambridge, England: Cambridge University Press.

Lawrence, G. (1993). *People types and tiger stripes: A practical guide to learning styles* (3rded.). Gainesville, FL: Center for Applications of Psychological Type.

Livesay, G.A., Dee, K. C., Nauman, E. A., & Hites, Jr. L. S. (2002). Engineering Student Learning Styles: A Statistical Analysis Using Felder's Index of Learning Styles, Presented at the 2002 Annual Conference of the ASEE, Montreal, Quebec, June 2002.

Lowe, R.K.(2003). Animation and learning: selective processing of information in dynamic graphics, *Learning and Instruction*, *13*, 157-176.

Ma, J., Nickerson, J. V. (2006). Hands-on, simulated, and remote laboratories: A comparative literature review. *ACM Computing Surveys.* 38 (3), 1–24.

Magin, D. J., Churches, A. E., Reizes, J. A. (1986). Design and experimentation in undergraduate mechanical engineering, *Proceedings of a Conference on Teaching Engineering Designers* (pp.96–100). Sydney, Australia: Institution of Engineers.

Markwell J., Brooks D. W. (2002). Broken Links: The Ephemeral Nature of Educational WWW Hyperlinks, *Journal of Science Education and Technology*, 11: pp. 105-108.

Marton, F., Säljö, R. (1997). Approaches to learning. In F. Marton, D. Hounsell, &N. Entwistle (Eds.), *The Experience of Learning* (2nd ed.). Edinburgh: Scottish Academic Press.

Mayer, R.E. (1993a). Multimedia learning: Are we asking the right question? *Educational Psychologist*, *32*(1), 1–19.

Mayer, R.E. (1993). Systematic thinking fostered by illustrations in scientific text, Journal *of Educational Psychology*, *81*(2), 240–246.

Mayer, R.E (2001). Multimedia learning. New York: Cambridge University Press.

Mayer, R.E. (2002). *The promise of educational psychology* (Vols. 1 & 2).Upper Saddle River, NJ: Prentice-Hall.

Mayer, R. E. (2003). The promise of multimedia learning: Using the same instructional design methods across different media, *Learning and Instruction*, *13*(12), 125-139.

Mayer, R. E. (2009). Multimedia learning (2nd ed.). New York: Cambridge.

Mayer, R. E. (2010). Unique contributions of eye-tracking research to the study of learning with graphics, *Learning and Instruction*, *20*, 167-171.

Mayer, R.E., Anderson, R.B. (1992). The instructive animation: Helping students build connections between words and pictures in multimedia learning, *Journal of Educational Psychology*, *84*(4), 444–452.

Mayer, R.E., Bove, W., Bryman, A., Mars, R., Tapangco, L. (1996). When less is more: Meaningful learning from visual and verbal summaries of science textbook lessons, *Journal of Educational Psychology*, *88*(1), 64–73.

Mayer, R.E., Gallini, J.K. (1990). When is an illustration worth ten thousand words? *Journal of Educational Psychology*, *82*(4), 715–726.

Mayer, R.E., Massa, L. J. (2003). Three facets of visual and verbal learners: Cognitive ability, cognitive style, and learning preference, *Journal of Educational Psychology*, *95*(4), 833–846.

Mayer, R.E., Moreno, R. (1998). A split-attention effect in multimedia learning: Evidence for dual processing system in working memory, *Journal of Educational Psychology*, *90*(2), 312–320.

Mayer, R.E., Moreno, R. (2002). Animation as an aid to multimedia learning, Educational *Psychological Review*, *14*(1), 87–99.

Montgomery, S.M. (1995). Addressing diverse learning styles through the use of multimedia, *Proceedings of the Frontiers in Education Conference*, November 11-14, 1995 Atlanta, Georgia, USA.

Teo, T. (2009).Modeling technology acceptance in education: a study of pre-service teachers, *Computers and Education*, 52(1), 302–312.

Moos, D., Azevedo, R. (2009). Learning with computer-based learning environments: A literature review of computer self-efficacy. *Review of Educational Research*, *79*(2), 576-600.

Najjar, L. (1998). Principles of educational multimedia user interface design, *Human Factors*, 40(2), 311-323.

Nersessian, N. J. (1991). Conceptual change in science and in science education. In M. R. Matthews (Ed.), *History, philosophy, and science teaching* (pp. 133-148). Toronto, Canada: OISE Press.

Nielsen, J. (1994). Heuristic evaluation. In Nielsen, J., and Mack, R.L., *Usability Inspection Methods*. JohnWiley & Sons, New York, NY.

Ong, C.-S., Lai, J.-Y., Wang, Y.-S. (2004). Factors affecting engineers' acceptance of asynchronous e-learning systems in high-tech companies, *Information and Management*, 41, 795–804.

Oviatt, S.L., Arthur, A.M., Cohen, J. (2006). Quiet interfaces that help students think. In *Proceedings of the Nineteenth Annual ACM Symposium on User Interface Software and Technology* Symposium conducted at the meeting of UIST '06. New York, NY: ACM Press.

Paas, F., Tuovinen, J., Tabbers, H., Van Gerven, P. (2003).Cognitive workload measurement as a means to advance cognitive workload theory. *Educational Psychologist*, *38*(1), 63-71.

Park, O. (1994). Dynamic visual displays in media-based instructions. *Educational Technology*, *21*.

Pashler, H. E. (1998). The psychology of attention. Cambridge, MA: MIT Press.

Paivio, A., and Simpson, H. M. (1966.) The effect of word abstractness and pleasantness on pupil size during an imagery task, *Psychometric Science*, 5, 55-56.

Payne, D. T., Parry, M. E. Harasymin, S. J. (1968).Percentage of pupillary dilation as a measure of item difficulty, *Perception and Psychophysics*, 4, 139-143.

Poole, A., and Ball, L. J. Eye tracking in human-computer interaction and usability research. In C. Ghaoui (ed.), *Encyclopedia of human computer interaction*. Idea Group, Pennsylvania, 2005, 211-219.

Pratt, R. W. (1970). Cognitive processing of uncertainty: Its effect on pupillary dilation and preference ratings. Perception and Psychophysics, 8, 193-198.

Reiber, L. P. (1991). Computer-based microworlds: A bridge between constructivism and direct instruction. *Annual Convention of the Association for Educational Communication and Technology* (pp. 692-707).

Rosen, Y. (2009). The effects of an animation-based online learning environment in transfer of knowledge and on motivation for science and technology learning. *Journal of Educational Computing Research*, *40*(4), 451-467.

Scanlon, E., Morris, E., Di Paolo, T., Cooper, M. (2002). Contemporary approaches to learning science: Technologically-mediated practical work. *Studies in Science Education*, *38*, 73–114.

Scheiter, K., Van Gog, T. (2009). Using eye tracking in applied research to study and stimulate the processing of information from multi-representational sources. *Applied Cognitive Psychology*, *23*, 1209-1214.

Simposon, H. M., Paivio, A. (1968). Effects on pupil size of manual and verbal indicators of cognitive task fulfillment. Perception and psychophysics, 3, 185-196.

Schmeck, R.R. (1998). *Learning strategies and learning styles*. New York, N.Y.: Plenum Press.

Schnotz, W., Lowe, R. K. (2003). External and internal representations in multimedia learning. *Learning and Instruction*, *13*(2), 117-123.

Shirley, T., and Todd, P. A. (1995). Understanding information technology USAGE: A test of competing models, *Information Systems Research*, *6*(2), 144-176.

Shneiderman, B., Zhang, P., and Galletta, D. (2006). *Human computer interaction and management information systems: Foundations. In Advances in Management Series.* SIGCHI (2001).*Notes from E-learning Special Interest Group (SIG) Discussion*, CHI.

Smith, P. J., Murphy, K. L., & Mahoney, S. E. (2003). Towards identifying factors underlying readiness for online learning: an exploratory study, *Distance Education.*, 24(1), 57–67.

Stice, J.E. (1987). Using Kolb's learning cycle to improve student learning. *Journal of Engineering Education*, 77(5), 291–296.

Street, S., Goodman, A. (1998) Some Experimental Evidence on the Educational Value of Interactive Java Applets in Web-Based Tutorials. In *Proceedings of the 3rd Australasian conference on Computer science education*.

Streufert, S.,Nogami, G. Y. (1989). Cognitive style and complexity: Implications for I/O psychology.In C.L. Cooper & I.Robertson (Eds.), *International Review of Industrial Design and Organizational Psychology*. Chichester: Wiley.

Teo, T. (2009). Modelling technology acceptance in education: a study of pre-service teachers, *Computers and Education*, 52(1), 302–312.

Triona, L. M., Klahr, D. (2003). Point and click or grab and heft: Comparing the influence of physical and virtual instructional materials on elementary school students' ability to design experiments. *Cognition and Instruction*, *21*(2), 149-173.

Tsianos, N., Germanakos, P., Mourlas, C., Meimaris, M. (2005). The learning styles as a basic parameter for the design of adaptive e-learning environments. *Proceedings of the 3rd International Conference on Open and Distance Learning*. Applications of Pedagogy and Technology: 651-668.

Venkatesh, V., Morris, M.G., Davis, G.B., Davis, F.D., (2003). User acceptance of information technology: toward a unified view. MIS Quarterly 27 (3), 425–478.

Yi, M. Y., Hwang, Y. (2003). Predicting the use of web-based information systems: selfefficacy, enjoyment, learning goal orientation, and the technology adoption model.*International Journal of Human Computer Studies*, 59(4), 431–449. Young, M. S., Stanton. N .A (2004). Taking the load off: Investigations of how adaptive cruise control affects mental workload, *Ergonomics*, 47(9), 1014-1035.

Wright, P., Kahneman, D. (1971) Evidence for alternative strategies of sentence retention, *Quarterly Journal of Experimental Psychology*, 23, 197-213.

Zaharias, P. (2005). E-learning design quality: A holistic conceptual framework. In *Encyclopedia of Distance Learning* (Vol.II, pp). Hershey: Idea Group Inc.

VITA

Arun Chintalapati was born in India on November 27, 1985. In May 2007 received his B.E in Information Technology from Osmania University in Hyderabad, India. In August 2007, he travelled to Seoul, South Korea for the Microsoft sponsored Imagine Cup competition. His team was selected from among thousands of other teams to participate in the final round.

He later pursued his Master of Science Degree at Missouri University of Science and Technology, Rolla. At Missouri University of Science and Technology, he worked as a Graduate Research Assistant from May 2009 to June 2010 in the Department of Business and Information Technology. He spent most of his time evaluating the Rapid Development System and helping his peers on other projects.

He worked as a technical trainer and usability specialist at Tobii Technologies, Inc. from June 2010 to May 2011. During his co-op he mainly provided training on using eye tracking equipment to several companies and carried out usability studies.

He received his MS in Information Science and Technology in December 2011.